Check for updates

OPEN ACCESS

EDITED BY Hongbo Yuan, Hebei Agricultural University, China

REVIEWED BY

Muhammad Shahid Iqbal, Anhui University, China Poonam Singh, LaserLeap Technologies, Portugal Joel Nwakaire, African Technology Policy Studies Network, Kenya

*CORRESPONDENCE Dongyi Wang,

☑ dongyiw@uark.edu
RECEIVED 30 May 2024
ACCEPTED 30 September 20

ACCEPTED 30 September 2024 PUBLISHED 17 October 2024

CITATION

Mahmoudi S, Davar A, Sohrabipour P, Bist RB, Tao Y and Wang D (2024) Leveraging imitation learning in agricultural robotics: a comprehensive survey and comparative analysis. *Front. Robot. Al* 11:1441312. doi: 10.3389/frobt.2024.1441312

COPYRIGHT

© 2024 Mahmoudi, Davar, Sohrabipour, Bist, Tao and Wang. This is an open-access article distributed under the terms of the Creative Commons Attribution License (CC BY). The use, distribution or reproduction in other forums is permitted, provided the original author(s) and the copyright owner(s) are credited and that the original publication in this journal is cited, in accordance with accepted academic practice. No use, distribution or reproduction is permitted which does not comply with these terms.

Leveraging imitation learning in agricultural robotics: a comprehensive survey and comparative analysis

Siavash Mahmoudi¹, Amirreza Davar^{1,2}, Pouya Sohrabipour¹, Ramesh Bahadur Bist¹, Yang Tao³ and Dongyi Wang^{1,4,5}*

¹Department of Biological and Agricultural Engineering, University of Arkansas, Fayetteville, AR, United States, ²Department of Mechanical Engineering, University of Arkansas, Fayetteville, AR, United States, ³Department of Bioengineering, University of Maryland, College Park, MD, United States, ⁴Department of Food Science, University of Arkansas, Fayetteville, AR, United States, ⁵Center of Scalable and Intelligent Automation in Poultry Processing, Fayetteville, AR, United States

Imitation learning (IL), a burgeoning frontier in machine learning, holds immense promise across diverse domains. In recent years, its integration into robotics has sparked significant interest, offering substantial advancements in autonomous control processes. This paper presents an exhaustive insight focusing on the implementation of imitation learning techniques in agricultural robotics. The survey rigorously examines varied research endeavors utilizing imitation learning to address pivotal agricultural challenges. Methodologically, this survey comprehensively investigates multifaceted aspects of imitation learning applications in agricultural robotics. The survey encompasses the identification of agricultural tasks that can potentially be addressed through imitation learning, detailed analysis of specific models and frameworks, and a thorough assessment of performance metrics employed in the surveyed studies. Additionally, it includes a comparative analysis between imitation learning techniques and conventional control methodologies in the realm of robotics. The findings derived from this survey unveil profound insights into the applications of imitation learning in agricultural robotics. These methods are highlighted for their potential to significantly improve task execution in dynamic and high-dimensional action spaces prevalent in agricultural settings, such as precision farming. Despite promising advancements, the survey discusses considerable challenges in data quality, environmental variability, and computational constraints that IL must overcome. The survey also addresses the ethical and social implications of implementing such technologies, emphasizing the need for robust policy frameworks to manage the societal impacts of automation. These findings hold substantial implications, showcasing the potential of imitation learning to revolutionize processes in agricultural robotics. This research significantly contributes to envisioning innovative applications and tools within the agricultural robotics domain, promising heightened productivity and efficiency in robotic agricultural systems. It underscores the potential for remarkable enhancements in various agricultural processes, signaling a

transformative trajectory for the sector, particularly in the realm of robotics and autonomous systems.

KEYWORDS

imitation learning, robotics, agricultural robotics, artificial intelligence, agricultural engineering

1 Introduction

In an era marked by demographic shifts and escalating food security concerns, the agricultural sector requires transformative solutions to meet increasing demands. The integration of advanced technologies, notably agricultural robotics, has received considerable attention and investment, evident from the growing scholarly focus on these innovations (Cheng et al., 2023). The constraints of traditional farming, marked by labor-intensive processes, imprecise environmental information measurement, and ineffective crop monitoring, underscore the pressing need for more streamlined and sophisticated agricultural practices leveraging artificial intelligence (AI) (Wang, 2019) and robotics (Yépez-Ponce et al., 2023). Researchers are actively working towards creating intelligent, cost-effective, and highly productive agricultural systems that integrate sensor technology and Internet of Things (IoT) (Brewster et al., 2017), data management (Wolfert et al., 2017), decision-making algorithms (Robert et al., 2016), robotics (Fountas et al., 2020), and advanced mechanisms to revolutionize traditional agricultural methods (Wakchaure et al., 2023). This push for innovation is not merely technological but deeply rooted in the urgent global need to enhance agricultural sustainability and food security, reflecting the critical role of agriculture in climate change and its socioeconomic impacts (Grieve et al., 2019). As such, the integration of robotic manipulations into agricultural practices represents a transformative leap, poised to optimize farming methodologies, augment productivity, and address the evolving needs of sustainable food production in the face of growing global demands (Grieve et al., 2019).

In recent years, from the foundational robotic research perspective, the integration of AI in robotic manipulation has undergone a paradigm shift, moving away from traditional predetermined control algorithms towards adaptive, learning-based control methodologies. This evolution, as noted by Yang et al. (2016) and Hussein et al. (2017), reflects the limitations of conventional control algorithm methods, particularly in enabling robots to dynamically adapt to varying tasks and environments. As we delve into the realm of agriculture, the challenges of integration of robotic technology presented are particularly noteworthy, primarily stemming from the inherent variability characterizing agricultural products and unconstrained field conditions. In contrast to industries marked by product standardization, the agricultural domain exhibits substantial diversity in terms of product morphology, including but not limited to variant size, visual features, and tactile textures (Eizicovits and Berman, 2014). The diversity necessitates the development of robotic systems capable of adapting to the intricate variations prevalent in agricultural environments, particularly in tasks such as planting, harvesting, and weeding (Gonzalez-de Santos et al., 2020). These challenges have provided excellent scenarios for the application of learning-based robotic control methodologies.

The general agricultural robotic systems, like other robotic systems, are composed of sensing systems, actuating systems, and control algorithms. For environmental sensing, visual recognition, plays an important role in recognizing and localization of objects of interest while presenting a formidable challenge, necessitating advanced systems capable of discerning subtle differences between the targeted objects and other environmental elements. The considerable variability in color, shape, and size among agricultural products demands the integration of sophisticated visual recognition technologies (Wu et al., 2021). Current research in this domain mainly leverages various Deep Learning (DL) models to enhance the accuracy of visual perception in robotic systems. Convolutional Neural Networks (CNNs), for instance, have been instrumental in processing and analyzing complex visual data, enabling robots to effectively differentiate between various plant species and detect anomalies such as disease or pest infestation (Prakash and Prakasam, 2023; Rezk et al., 2022). Techniques such as transfer learning have also been applied, allowing models trained on extensive datasets to be adapted for specific agricultural tasks by fine-tuning based on relatively small task-specific datasets, thereby improving efficiency and reducing the need for extensive field-specific data collection (Goel et al., 2022; Moiz et al., 2022). In addition to visual sensing, the tactile sensing capabilities equipped by robotic systems enable the robots to potentially handle agricultural products ranging from texture to firmness which pose significant hurdles in the practice, such as for the robotic harvesting tasks (Mandil et al., 2023). Recent advancements have seen the development of various types of tactile sensors, including capacitive, resistive, and piezoelectric sensors, each offering unique benefits in terms of sensitivity and adaptability to different materials and surfaces (Bayer, 2022; Baldini et al., 2023). For example, capacitive sensors, which measure changes in capacitance when in contact with an object, have been effectively used in robotic grippers to ascertain the firmness and ripeness of fruits, enabling delicate handling and minimizing damage (Alshawabkeh et al., 2023).

In practice, addressing the comprehensive challenges in agriculture, such as unpredictable environmental conditions and the need for precise and gentle handling of process like chicken rehanging tasks in the poultry industry, requires a holistic approach in robotic system design. Robotic pollination is also an important consideration for similar environments, where the need for adaptability and precision in delicate tasks, like pollinating crops in large greenhouses, mirrors the intricacies seen in other agricultural operations. (Broussard et al., 2023). As bee populations decline, robots like the PollyPollinator are being developed to autonomously pollinate crops, particularly in large greenhouses (Ohi et al., 2018). Automated greenhouse management is also advancing through the use of imitation learning techniques. Robots like Sweeper are used

to autonomously harvest bell peppers, demonstrating the ability to perform delicate tasks in controlled environments (Arad et al., 2020). These systems ensure that optimal growing conditions are maintained while minimizing energy consumption, showcasing the potential of robotics to improve efficiency and sustainability in agriculture. Mentioned tasks, characterized by their complexity and the need for fine motor skills and judgment, are difficult to articulate concisely and standardize across varying conditions. Integrating of multi-sensory data, combining visual and tactile feedback with machine learning algorithms, is a growing area of focus. This multi-sensory integration allows for more nuanced decision-making processes in robots and enhanced dexterity and cognitive abilities, making them capable of performing tasks that traditionally rely heavily on human expertise and adaptability.

Beyond arm manipulation, robotic control critically encompasses the dynamic movement of robotic vehicles. The agricultural landscape, with its uneven and unpredictable terrain, poses significant navigation challenges that traditional wheeled or tracked robots might not efficiently overcome. These challenges necessitate the development of robots with enhanced mobility capabilities to navigate diverse topographies, including slopes, rough surfaces, and areas with soft soil that could impede traditional means of locomotion (Botta et al., 2022). In response to these challenges, legged robotic control emerges as a promising solution, offering superior adaptability and mobility in complex agricultural environments. Legged robots, inspired by the locomotion of animals, can traverse obstacles, step over gaps, and adjust their body configuration to maintain stability on uneven terrain (Yang et al., 2023). This adaptability is crucial in agriculture, where the robots must operate in fields with variable soil types, undulating surfaces, and around crops planted in irregular patterns. Recent advancements in legged robotics, fueled by sophisticated control algorithms and sensory feedback systems, have significantly improved their efficiency and robustness. Techniques such as reinforcement learning and bio-inspired control strategies have been applied to optimize legged locomotion, enabling these robots to make real-time decisions based on environmental feedback. This allows for precise movement control, essential for tasks such as targeted spraying, soil analysis, and crop monitoring, minimizing the risk of damaging the crops or soil structure (Kaur and Bawa, 2022; Sun et al., 2023).

Traditional control strategies in robotics, such as PID (Proportional-Integral-Derivative) controllers and model predictive control (MPC), have been instrumental in enabling precise and reliable actions in structured environments. However, as mentioned above, the agricultural domain presents unique challenges that often exceed the capabilities of these traditional approaches (Li et al., 2023). The inherent variability of natural environments, coupled with the need for delicate handling of agrifood products and navigation through unstructured terrains, calls for more advanced and flexible control strategies.

As these challenges are intricately interconnected, The development of robotic solutions for agriculture epitomizes the quintessence of interdisciplinary collaboration, requiring a confluence of expertise from robotics, computer vision, AI, and agricultural sciences. Such a coalition is indispensable for engendering systems that are not merely adaptable and robust but also keenly attuned to the multifarious demands of agriculture.

This is where Imitation Learning (IL) emerges as a particularly valuable tool mimicking expert human behaviors, IL enables robots to assimilate complex tasks with remarkable efficiency, significantly curtailing the need for extensive programming (Yamin and Bhat, 2022). The indispensability of IL in agricultural robotics is underscored by its ability to circumvent the limitations of traditional control mechanisms. IL facilitates the transfer of human expertise and human-like dexterity to robotic systems, empowering them to perform intricate tasks such as precision planting, targeted pesticide application, selective harvesting, and, notably, complex food processing operations. This approach is instrumental in enhancing the adaptability of robots to dynamic agricultural and food processing environments, where precision and delicacy are paramount. By optimizing resource utilization and minimizing environmental impact, IL-based robotic systems represent a promising strategy for enhancing productivity and sustainability in the agriculture and food processing sectors (Krithiga et al., 2017). Furthermore, the integration of IL within agricultural robotics exemplifies the synergistic potential of combining machine learning with other technological advancements in AI and computer vision which paves the way for enlarging the application scope of the advanced robotic techniques in precision agriculture (Daaboul et al., 2019).

This paper explores the application of IL within the domain of agricultural robotics, showcasing how the synergistic integration of robotics, AI, and agricultural sciences—enhanced through IL—signifies a revolutionary shift in the field. It emphasizes the critical need for evolving the broader landscape of IL research to address the intricate challenges faced by the agricultural sector, thereby enabling robotic systems to play a pivotal role in the advancement of sustainable agriculture.

Hussein et al. (2017) present a seminal survey that systematically categorizes various IL methods, detailing the diversity of learning strategies deployed across multiple domains. This comprehensive review is particularly invaluable as it traces the evolution of IL, outlining key methodological advancements and conceptual frameworks that have shaped the field. Their discussion extends beyond mere categorization, providing critical insights into how different approaches address specific computational challenges, thereby serving as a foundational reference for understanding the progression and refinement of IL technologies.

In a more recent survey, Zare et al. (2024) offer a forwardlooking perspective by examining the latest algorithmic advancements and identifying persisting challenges within IL up to the year 2024. Their survey not only synthesizes recent developments but also emphasizes the application of IL in dynamically complex environments. Their findings are especially relevant to agriculture, where IL must adapt to variable and unpredictable conditions, thus highlighting potential areas for technological interventions to enhance precision farming practices. Torabi et al. (2019) delve into the subfield of IL from observation, which becomes crucial in scenarios where explicit action data is limited or absent. This aspect of IL is highly pertinent to agriculture, where capturing detailed action data can be challenging. The review paper explores stateonly learning approaches, discussing their practical applications and theoretical underpinnings. The adaptability of these methods to agricultural settings offers significant potential for developing

autonomous systems that learn from observational data alone, thus simplifying the integration of IL in field operations.

In more specialized application of IL, Le Mero et al. (2022) review IL techniques for autonomous vehicles, focusing on endto-end learning for autonomous navigation systems. Although their primary context is vehicular automation, the methodologies they discuss are adaptable to autonomous agricultural machinery, potentially enhancing efficiency in field operations. Fang et al. (2019) delve into robotic manipulation, an area that involves acquiring fine motor skills from human demonstrations. Their insights are particularly applicable to robotic systems that might be employed in precision agriculture tasks such as planting or harvesting.

While the reviews mentioned above offer an extensive overview of imitation learning across diverse disciplines and technical dimensions, it's essential to underscore the unique orientation of our work. Moving beyond the technical intricacies, our review delves into how these methodologies can be specifically adapted and applied to agricultural contexts. This tailored approach isn't just about exploring theory; it involves a deep dive into practical applications that are directly relevant to the challenges and needs of the agricultural sector. By bridging theoretical insights with realworld practices, our review aims to provide a practical toolkit for agricultural practitioners and innovators, helping them to harness these advanced techniques in ways that are both effective and meaningful for their specific fieldwork.

The remainder of the paper is organized as follows: Section 2 outlines the search criteria employed to identify relevant articles for this review, detailing both the inclusion and exclusion parameters. Section 3 provides an overview of IL, explaining the key concepts, algorithms, and their specific adaptations for agricultural applications. In Section 4, case studies and practical implementations have been presented that illustrate the successful application of IL in various agricultural settings. Section 5 addresses the challenges and limitations of current IL approaches, offering a critical analysis of what is needed to overcome these barriers. Finally, Section 6 outlines the future directions for research in IL within agriculture, proposing potential innovations and improvements that could further enhance the efficiency and effectiveness of robotic systems in this vital sector.

2 Methodology of literature selection

A thorough literature search was conducted across three prominent databases: Web of Science, Science Direct, and Google Scholar, covering studies published from 1985 to 2024. The search focused on imitation learning and its application in agriculture. Many research papers on imitation learning models are available, but no critical review paper is available based on agriculture; therefore, this review holds significant importance. It was searched using special keywords like imitation learning, imitation learning in agriculture, agricultural robotics, precision agriculture, applied artificial intelligence, and smart agriculture. Boolean operators ("AND" and "OR") were used to refine and narrow the search results, ensuring the retrieved literature was highly relevant. Each abstract was meticulously reviewed to align with the study's objectives, which centered on two key areas within imitation learning models:



data collection and real-time applications. The search was tailored to include papers that employed imitation learning models in agricultural contexts or provided methodological insights into these models.

Our review prioritized papers utilizing imitation learning models to achieve real-time learning applications or develop efficient data collection. This paper employed automation tools within the database search engines to streamline and filter the search results, ensuring a rigorous and efficient search process. Only studies published in English were considered, while conference papers and literature from unrelated domains were excluded to maintain focus. In total, 59 research articles were selected for inclusion in this systematic review, providing a comprehensive overview of current advances and applications of imitation learning in agriculture. A review of the existing literature followed the guidelines set by the Preferred Reporting Items for Systematic Reviews and Meta-Analyses (PRISMA) (Page et al., 2021) and is self-explained in Figure 1.

3 Foundations of imitation learning

This section provides a comprehensive overview of IL, situating it within the broader context of machine learning and robotics. It emphasizes IL's distinctive methodology, which allows robotic systems to acquire complex behaviors by observing expert practitioners. Our objective is to explore in depth the fundamental aspects of imitation learning, examining its principal methodologies and principles. We will also discuss the significant impact these techniques have on the advancement of agricultural robotics. A thorough analysis of various imitation learning strategies will reveal how these methods enable robots to learn skills effectively, thereby allowing them to execute intricate agricultural operations with enhanced precision and efficiency.

3.1 Introduction to imitation learning

IL introduces a pivotal development in the fields of machine learning and robotics, characterized by its unique method of behavioral acquisition from expert demonstrations. Unlike traditional supervised learning, which depends on discrete labels, or reinforcement learning, which is founded on trial-and-error, IL enables robots to learn complex behaviors by closely observing human expertise (Argall and Billard, 2010). This methodological advancement is essential for applications that require adaptability and precision, particularly in dynamic environments where traditional control methods are inadequate (Schaal, 1999).

The evolution of IL from its conceptual inception to its current applications in advanced robotics showcases the shift from simpler methods to increasingly complex strategies. Initially, IL was primarily about simple behavior mimicry, as demonstrated in Pomerleau's work (1988), where robots learned actions directly from expert demonstrations without understanding the underlying intentions or rewards. Recent advancements include sophisticated models like Generative Adversarial Imitation Learning (GAIL). In GAIL, a generator (the imitating agent) learns to mimic actions indistinguishably from an expert, under the scrutiny of an adversarially trained discriminator (Ho and Ermon, 2016). Building on the principles of GAIL, newer methods, such as diffusion policy models have emerged. These models integrate various learning techniques to enhance the adaptability and efficiency of IL, enabling robots to perform in more dynamic and unpredictable environments by effectively merging multiple policies into a coherent strategy (Chi et al., 2023). This progression underscores the growing significance and complexity of IL in the field of advanced robotics. Figure 2 shows a simple flow from expert input data to the agent's learning process, indicating the interactions between state, observation, and action, as guided by an expert's demonstrated policies.

IL encompasses a range of strategies essential for advancing robotics, notably Behavioral Cloning (BC), Inverse Reinforcement Learning (IRL), and Imitation from Observation (IfO). In Figure 3, a comprehensive visual comparison of imitation learning methods is presented by categorizing them into two groups: primary methods and advanced methods tailored for agricultural applications. The figure highlights each method's respective process, starting from data collection to policy deployment, while also providing detailed insights into the unique features of each method.

BC involves the direct replication of behavior by learning from expert actions in specific states, while IRL focuses on deriving underlying reward functions from observed expert behaviors, aiming to understand and replicate the motivations behind these actions. IfO allows for learning without access to expert actions, instead relying solely on observing the state transitions made by the expert. Additionally, advanced methods such as Generative Adversarial Imitation Learning (GAIL) and Diffusion Policy expand the toolkit available for tackling unique challenges in robotics applications, particularly in agriculture. GAIL utilizes adversarial processes to train policies that can not only mimic but also generalize from expert behavior, whereas Diffusion Policy integrates multiple policy models or data sources to create a robust consensus policy, enhancing performance in complex agricultural environments. These advanced techniques help solidify imitation learning as a cornerstone of modern robotics innovation, especially in domains requiring sophisticated interaction with the environment.

3.2 Primary methods

3.2.1 Behavioral cloning

The transformative potential of IL in agriculture is its capability to introduce automation and data-driven decision-making into traditional farming practices. BC stands out for its direct and efficient approach, excelling in learning and replicating expert behaviors through a method akin to supervised learning. This method entails constructing a predictive model that maps environmental states to corresponding expert actions, utilizing a dataset of state-action pairs meticulously documented from expert performances. In practice, states can be described by various factors, including soil moisture levels, plant growth stages, or weather conditions, providing a comprehensive context for the robot's actions. These pairs serve as a model blueprint, guiding it to emulate expert decision-making processes under the assumption that a sufficiently diverse and comprehensive dataset can enable the model to generalize well to new, unseen states (Pomerleau, 1988; Ross et al., 2011). BC's inherent simplicity, deriving from its problem space abstraction, liberates it from the complexities of environmental dynamics. This characteristic grants BC high adaptability and precision, allowing for its application across a wide range of agricultural tasks by learning from visual and tactile cues and corresponding human expert demonstrations.

The advantage of BC in agriculture lies in its capacity to function independently of complex environmental intricacies. This facilitates its deployment in a variety of crucial agricultural operations. For example, in precision irrigation, BC enables robots to learn from expert demonstrations on when and where to irrigate, considering factors like soil moisture content and weather predictions. By observing the timing, amount, and methods used by experts in different conditions, robots can make data-driven irrigation decisions, optimizing water usage for crop health and yield without constant human supervision. Similarly, in autonomous pest control, BC can teach robots to identify and respond to pest infestations based on visual cues and expert interventions. Through learning from experts' actions, such as the application of pesticides or the removal of infected plants, robots can autonomously navigate fields, detect pest activity, and apply precise interventions. This not only increases the efficiency and effectiveness of pest control measures but also reduces the reliance on manual labor and the potential for human error.



Both applications exemplify how BC, by distilling complex decision-making processes into learnable patterns from expert demonstrations, contributes significantly to the enhancement of modern farming operations. These implementations of BC in agriculture align with the sector's growing need for real-time, data-driven decision-making, underscoring the methodology's adaptability and potential to revolutionize traditional farming practices (Kim et al., 2023; Zhang C. et al., 2023).

The objective central to BC is to learn a policy π that emulates expert behavior by minimizing the discrepancy between predicted and expert actions in identical states, formalized in Equation 1:

$$\min_{\pi} \frac{1}{N} \sum_{i=1}^{N} \ell(\pi(s_i), a_i)$$
(1)

Here, *N* represents the total number of state-action pairs from expert demonstrations, where each pair consists of a state s_i and the corresponding action a_i taken by the expert. The action a_i can range from simple binary decisions (e.g., turn on or off the irrigation) to complex continuous actions (e.g., adjusting the amount of pesticide to distribute based on pest density). The loss function $\ell(\pi(s_i), a_i)$ quantifies the discrepancy between the policy's predicted action $\pi(s_i)$ and the expert's actual action a_i for each state. A commonly used loss function in BC is the mean squared error (MSE) for continuous operations which have a series of state-action pairs. MSE calculates the square of the difference between predicted and actual actions. For categorical actions, a cross-entropy loss function might be used, measuring the dissimilarity between the predicted probability distribution over actions and the distribution representing the expert's action. By effectively minimizing this loss, BC enables the model to closely approximate expert behavior given a specific environmental state, thereby facilitating a direct learning approach from expert demonstrations (Pomerleau, 1988). This process underscores BC's practicality and efficiency in transmitting human expertise to robots, enhancing their capability to perform complex agricultural tasks with a level of precision and adaptability that mirrors human performance.

Despite BC's advantages in automating complex tasks by mimicking human behaviors, it confronts the covariate shift problem, a significant challenge when transitioning from training to deployment in more general practical settings. The BC model, trained under expert-generated states such as those from simulated environments, controlled settings, or limited training data, struggles to adequately perform or recognize out-of-distribution states in real agricultural environments. These discrepancies can lead to safety and precision issues, as the model may not have been exposed to the full range of real-world variations during its training phase (Zhou K. et al., 2022; Reddy et al., 2019).

Addressing the covariate shift in BC requires exploring strategies beyond conventional BC methods. These strategies include Interactive Imitation Learning (IIL), expert policy support estimation, and the development of constrained operational domains (Chang et al., 2021). Each strategy aims to overcome BC's inherent limitations, promoting the deployment of robust and adaptable models suitable for the dynamic agricultural environment.

IIL, with Dataset Aggregation (DAgger) as a pioneering method, minimizes discrepancies between training and testing environments



by integrating expert feedback into the learning process, enabling adaptive learning that gradually incorporates real-world feedback (Ross et al., 2011; Ross and Bagnell, 2014). This method can be described in Equation 2 as:

$$\pi_{\mathrm{DAgger}} = \arg\min_{\pi} \sum_{i=1}^{N} \mathbb{E}_{(s,a) \sim \pi_i}[\ell(\pi(s), a)]$$
(2)

The equation represents the optimization process to find the best policy (π), which minimizes the expected loss (ℓ) over a series of iterations (N). Here, $\mathbb{E}_{(s,a)\sim\pi_i}[\ell(\pi(s),a)]$ calculates the expected loss of actions (a) taken in states (s) as predicted by the current iteration of the policy (π), compared to the expert actions. This approach underscores the model's continuous adjustment and learning from the discrepancies between its actions and those of the expert, aiming to reduce these differences over time. However, this method introduces challenges, notably the increased cognitive load on human experts, who must provide continuous feedback to refine the model's accuracy and effectiveness. The requirement for sustained expert involvement can be demanding and may necessitate strategies to manage this aspect efficiently (Li et al., 2022; Kelly et al., 2019; Zhang and Cho, 2017; Hoque et al., 2021).

The estimation of expert policy's support focuses on defining reward structures that encourage the emulation of expert behavior, which is crucial for precision agricultural tasks (Reddy et al., 2019). Meanwhile, constrained operational domains can be formalized in Equation 3 to ensure that agricultural machinery operates within parameters well-represented in the training data, minimizing the risk of encountering unpredictable states (Dadashi et al., 2020):

$$\min_{\pi} L(\pi) \quad \text{subject to} \quad s \in S_{safe} \tag{3}$$

Here, $L(\pi)$ represents the loss function for the policy π , and S_{safe} denotes the set of states considered safe or well-represented in the training data. By constraining the robot to operate only within these predefined safe states, we significantly reduce the likelihood of encountering unpredictable situations that could lead to errors or inefficiencies, thereby enhancing the overall reliability and safety of agricultural operations (Mitka, 2018). This comprehensive approach, supported by a robust theoretical foundation and empirical validation, enhances BC models to suit the complex and variable nature of agricultural environments, ensuring efficiency, safety, and operational fidelity. In addressing the covariate shift challenge, particularly pertinent to applications like autonomous tractors and drones in agriculture, the augmentation of the imitation loss with additional constraints has proven effective (Bansal et al., 2018). Incorporating synthetic perturbations into the expert's trajectory exposes the model to non-expert behaviors, such as potential near-collision scenarios, which is crucial for the model to learn avoidance behaviors, thereby enhancing operational safety in agricultural tasks. The augmented loss function is given by Equation 4 as below:

$$L_{\text{total}} = L_{\text{imitation}} + \lambda (L_{\text{safety}} + L_{\text{perturbation}})$$
(4)

Here, $L_{\rm imitation}$ denotes the standard imitation loss, focusing on replicating the expert's actions. $L_{\rm safety}$ penalizes actions leading to unsafe outcomes, and $L_{\rm perturbation}$ introduces controlled deviations from the expert's trajectory to simulate unexpected scenarios. The weighting factor λ plays a critical role in balancing these contributions, ensuring that the model's training is comprehensive and geared towards operational safety in agricultural contexts.

Learned error detection systems enhance the safety of autonomous agricultural robots by ensuring operations stay within previously demonstrated safe limits. These systems identify potential failure states, restricting actions to prevent errors in essential tasks like crop monitoring and pest control (Wong et al., 2022; Spykman et al., 2021). For crop monitoring, safety constraints might prevent contact with plants to avoid damage. In pest control, constraints ensure precise pesticide application to protect non-target plants and beneficial insects. Such systems proactively mitigate risks by limiting robot actions based on learned safe behaviors, thus safeguarding both crops and the environment.

To enhance the adaptability of BC models in agricultural robotics, it is crucial to address the challenges of learning in the field. Agriculture involves diverse tasks and environmental conditions, necessitating rapid learning from minimal examples. However, comprehensive human demonstrations for training are often constrained by the availability and variability of data, which will be discussed in detail in Section 4. Techniques like Model-Agnostic Meta-Learning (MAML), offline imitation learning, and implicit BC models are vital for enabling robots to adapt quickly, learn from varied datasets, and perform effectively in complex agricultural environments.

MAML emerges as a solution, significantly enhancing a robot's adaptability with its efficient approach to learning from a few instances. MAML's optimization formula (Equation 5) is key to its success:

$$\theta' = \theta - \alpha \nabla_{\theta} L_T(f_{\theta}) \tag{5}$$

In this formula, θ' are the model parameters post-update, θ are the initial parameters, α is the learning rate, and L_T represents the task-specific loss function. MAML begins by training a general model across a variety of tasks to learn an initial set of parameters. These parameters are then fine-tuned for specific tasks through one or a few gradient descent steps, allowing the model to adapt quickly to new tasks from minimal data. MAML's capacity for quick adaptation from minimal data proves invaluable in agricultural settings, where operational conditions and task requirements can rapidly change due to environmental factors (Finn et al., 2017). Despite its computational demands, MAML's potential to facilitate swift adaptation with few examples aligns perfectly with the needs of the agricultural sector. This approach equips robotic systems with the flexibility required for precise monitoring and management in diverse agricultural scenarios, reinforcing the essential role of advanced learning techniques in evolving farming practices (Grant et al., 2018).

Offline imitation learning stands out as another strategic approach to bolstering the robustness of models, especially in contexts where engaging directly with experts is challenging. This method is instrumental in addressing the scarcity of data by leveraging a combination of precise expert policy stateaction pairs alongside expansive datasets featuring less-than-ideal or suboptimal behaviors. Such a blend enriches the training material, presenting both ideal scenarios and common errors or variations (Zhang W. et al., 2023). The core of offline imitation learning lies in refining a dynamics model, which is taught to recognize and correct deviations from desired actions by imposing increased penalties on inaccurately represented segments of the state-action space (Chang et al., 2021). For agricultural applications, where diverse conditions and unpredictable variables make collecting comprehensive and optimal practice data difficult, offline imitation learning proves particularly useful for handling diverse and suboptimal datasets. For example, Liu et al. (2021) propose a curriculum offline imitation learning strategy that mitigates the drawbacks of mixed-quality datasets by using an adaptive experience-picking strategy to improve policy learning. Similarly, DeMoss et al. (2023) introduce DITTO, an algorithm that uses world models to address covariate shifts without additional online interactions, which is crucial for applications in unpredictable environments like agriculture. Moreover, Zhang W. et al. (2023) present a discriminator-guided model-based offline imitation learning framework that enhances performance by distinguishing between expert and suboptimal data, improving model robustness. Additionally, Park & Yang (2022) discuss the benefits of leveraging suboptimal datasets alongside optimal data, demonstrating that this approach can significantly improve learning outcomes even when expert data is scarce. By incorporating these insights, offline imitation learning proves to be a highly effective strategy for training agricultural robots, making them more adaptable and reliable in real-world scenarios.

Finally, the development of implicit BC models can also address the need for decisiveness in agricultural robots, which is essential for tasks requiring transitions between different types of crops or soil conditions. By framing BC as an energy-based modeling problem, where the model outputs a value indicative of action appropriateness, these models are capable of representing discontinuities effectively (Florence et al., 2022):

$$E_{BC}(x,a) = -\log P(a|x) \tag{6}$$

Here, in Equation 6, E_{BC} signifies the energy associated with taking action *a* given observation *x*, with log P(a|x) representing the probability of the action being expert-like. Studies have shown that implicit behavioral cloning policies, particularly energy-based models, often outperform explicit models in various robotic policy learning tasks, including those with high-dimensional action spaces and visual image inputs (Balesni et al., 2023). These models

Strategy	Key characteristics	Advantages	Limitations	Agricultural applications	References
Interactive Imitation Learning (IIL)	Dynamically collects training data based on the model's performance. Integrates expert feedback iteratively	Continuous model improvement. Reduces training-testing gap	High expert dependency. Cognitive load on experts	Precision irrigation. Autonomous pest control	Ross et al., 2011; Finn et al., 2017
Expert Policy Support Estimation	Estimates the expert policy's support to guide learning. Uses reward shaping based on expert behavior	Enhances model fidelity. Tailorable to specific tasks	Requires deep expert behavior insight. Difficult to generalize	Crop monitoring. Yield optimization	Reddy et al., 2019
Constrained Operational Domains	Defines safe operational boundaries. Uses task-specific constraints	Increases safety and predictability. Reduces novel state risks	May restrict operational flexibility. Requires extensive domain knowledge	Ground vehicle navigation. Drone surveillance	Dadashi et al., 2020; Kim et al., 2023
Synthetic Data Augmentation	Uses artificial data to simulate out-of-distribution states. Prepares model for a wider range of scenarios	Enhances model robustness without real-world data. Tailorable to specific challenges	Risk of unrealistic data. Requires careful design	Adverse weather simulation. Rare pest outbreak training	Bansal et al., 2018; Wong et al., 2022
Implicit BC Models	Frames BC as an energy-based problem. Outputs a value indicating action appropriateness	Effective in representing discontinuities. Decisive in varied conditions	Complex model interpretation. Requires careful tuning	Crop type transitions. Soil condition adaptations	Florence et al. (2022)
Offline Imitation Learning	Utilizes both expert and suboptimal behavior data. Trains a dynamics model to apply penalties in poorly represented areas	Enhances model robustness in limited data scenarios. Mitigates overfitting to export data	Balancing expert and suboptimal data is challenging. Risk of learning from suboptimal actions	Data-scarce agricultural practices. Complex task learning	Chang et al. (2021)
Causal Misidentification	Identifies true causal relationships behind expert actions. Employs targeted interventions to refine accuracy	Improves model adaptability to dynamic conditions. Ensures training on the causal structure of tasks	Requires identification of accurate causal relationships. Potentially high complexity in modeling	Understanding crop growth factors. Pest control decision-making	De Stefano (2019)
Model-Agnostic Meta-Learning (MAML)	Optimizes model for quick adaptability to new tasks with minimal data	Rapid adaptation to varied conditions. Model-agnostic, broad applicability	Computationally intensive. Requires diverse training tasks	Rapid crop type adaptation. Environmental condition adjustments	Finn et al., 2017; Grant et al., 2018

TABLE 1 Comparative analysis of strategies to overcome covariate shift and improving BC model adaptability in agricultural application.

have demonstrated the ability to handle complex, discontinuous functions and improve policy learning in scenarios where traditional methods struggle (Singh et al., 2023).

In summary, incorporating a diverse array of strategies to mitigate covariate shifts and enhance model adaptability is crucial for advancing the field of agricultural robotics. Our comparative analysis, detailed in Table 1, showcases a spectrum of aforementioned methodologies, each offering unique benefits and challenges. This comparative insight is to guide future research and application development, ensuring the selection of strategies that not only address the inherent challenges of agricultural tasks but also leverage the full potential of behavioral cloning technologies.

3.2.2 Inverse reinforcement learning

Inverse Reinforcement Learning (IRL) marks a significant advancement in computational learning sciences, emphasizing the extraction of underlying objectives or "rewards" that guide expert decisions, beyond just mimicking observed actions as BC. This focus is particularly crucial in agriculture, where decision-making incorporates complex variables that standard models may overlook. IRL excels by deducing the reward functions that experts implicitly follow, providing deeper insights into their decision drivers. At its core, IRL aims to uncover the 'why' behind the 'what' by analyzing expert actions to discover the hidden rewards that dictate such behaviors. Contrasting with Behavioral Cloning, which directly maps observations to actions, IRL delves into replicating the decision-making process itself. This distinction is critical in situations where direct observation alone does not clearly suggest the best action to take (Ng and Russell 2000).

The primary focus of IRL is on the reward function,R(s), which assesses the desirability of a state *s*. In practical terms, especially within agricultural applications, the explicit reward function is hard to customize. As outlined in 7, the objective of IRL is to reverse-engineer this function from observed behaviors, effectively tackling the inverse of the usual reinforcement learning problem, where the goal is to learn the reward function that explains the observed behaviors (Arora and Doshi, 2021). Equation 7, highlited the mentioned approach.

Given: observed behaviors \rightarrow Find: R(s) that explains these behaviors (7)

In IRL, the observed behaviors are utilized to infer the underlying reward structure. For instance, Malik et al. (2018) introduced cooperative inverse reinforcement learning (CIRL), which formalizes the value alignment problem as a game between a human and a robot. In CIRL, the robot learns the human's reward function through interaction, leading to better alignment with human intentions. Also, Korsunsky et al. (2019) demonstrated the application of IRL in contextual Markov Decision Processes (CMDPs), where the reward function is inferred based on varying contexts. Their work highlighted the ability to generalize across different contexts by learning a mapping from contexts to rewards, which is particularly useful in dynamic agricultural settings where environmental conditions and tasks can vary significantly. Additionally, Lindner et al. (2022) proposed an active exploration strategy for IRL, which enhances the efficiency of learning the reward function by selectively querying the most informative regions of the state space. This method significantly improves the sample efficiency and the accuracy of the inferred reward functions, making it highly applicable to agricultural robotics where data collection can be expensive and time-consuming. By focusing on the underlying reward structures, IRL provides a more nuanced and flexible approach to understanding and replicating expert behaviors, thus enhancing its applicability in dynamic and complex environments (Shao and Er, 2012).

The scope of IRL's application extends to multi-agent settings, demonstrating its scalability and the ability to infer reward functions that accurately generalize across varied settings and interactions. This broadens the potential of IRL in complex, interactive environments. Tailored IRL algorithms for continuous state spaces that provide formal guarantees on sample and time complexity are crucial for the development of reliable, automated control systems in precision agriculture (Fu et al., 2021; Dexter et al., 2021). Moreover, the challenge of model misspecification in IRL, especially significant in complex human-involved environments, is being addressed through rigorous mathematical analysis. This enhances the robustness of IRL models, ensuring the reliability of inferences drawn from observed behaviors. The introduction of innovative approaches that transcend the limitations of generative models, encoding versatile behaviors through iteratively trained discriminators, showcases the exceptional generalization capabilities of IRL. Such advancements underscore the diversity of expert behaviors and strategies that can be captured and implemented in autonomous agricultural systems (Skalse and Abate, 2023; Freymuth et al., 2021).

3.2.3 Imitation from observation

IL typically requires both observing the states and the actions demonstrated by an expert. This dual requirement often necessitates comprehensive data collection, which can be restrictive and laborintensive, especially in complex environments like robotics and gaming (Edwards et al., 2019). Experts might need to operate robots directly or use specialized software for recording actions in gaming, which not only demands significant effort but also limits the data to artificial conditions (Hu et al., 2022). Imitation from Observation (IfO) simplifies this by learning policies from state transitions alone, eliminating the need for action data (Liu et al., 2018). This approach mirrors natural learning processes seen in humans and animals, who can learn complex behaviors just by watching. IfO, therefore, broadens the scope of IL, making it possible to utilize diverse and previously inaccessible resources such as online instructional videos (Aytar et al., 2018).

IfO discards the necessity for direct action observation and instead utilizes state-only demonstrations. It leverages advanced machine learning algorithms, including deep learning and computer vision, to parse and interpret raw visual data (Choe et al., 2021). The foundational model for IfO can be formalized in Equation 8 by considering a set of state transitions observed from an expert's demonstration:

$$s_{t+1} = f(s_t, a_t; \theta) \tag{8}$$

Where s_t and s_{t+1} are consecutive states, a_t is the expert's action which is typically unobserved in IfO, and θ represents the parameters of a model learned from data.

The key to IfO is the derivation of an optimal policy π that maximizes the expected reward over a trajectory, formalized under the reward function *R* derived from the state transitions as presented in Equation 9:

$$\pi^* = \arg\max_{\pi} \mathbb{E}\left[\sum_{t=1}^{T} R(s_t, \pi(s_t))\right]$$
(9)

In the IfO framework, consecutive states s_t and s_{t+1} are modeled by the transition function f with parameters θ , where the action a_t at time t, typically unobserved, influences the policy π that aims to maximize the cumulative reward over a trajectory from t = 1to T, calculated as the expected sum of rewards $R(s_t, \pi(s_t))$, which assess the suitability of the actions recommended by π at each state. Liu et al. introduced an approach that utilizes context-aware translation models to adapt the expert's demonstrations from an extrinsic viewpoint to the learner's intrinsic perspective (Liu et al., 2018). This method essentially involves a function f that translates the context of input observations s from the demonstration: $s' = f(s; \theta)$. Here, s' is the state as perceived in the learner's behavior effectively.

In agriculture, IfO has the potential to revolutionize various practices by automating complex tasks such as planting, weeding, and harvesting. Robots equipped with IfO capabilities can learn from videos showing human farmers performing these tasks, thereby adapting human dexterity and judgment to robotic systems. For instance, in automated crop harvesting, IfO can facilitate the development of robotic systems that observe and learn the optimal techniques for harvesting specific crops. This application involves not only recognizing crop maturity via visual cues—akin to the capabilities developed through TCN (Time-Contrastive Networks) methods (Sermanet et al., 2018)—but also mimicking the motion patterns used by human harvesters to minimize damage to both the produce and the plant.

Despite its potential, IfO faces significant challenges, particularly in terms of data quality and environmental variability. The successful implementation of IfO depends heavily on the availability of highquality, representative training data, which in many agricultural settings, must be captured across varying conditions to ensure robustness (Raychaudhuri et al., 2021). Additionally, the alignment of demonstrations across different contexts and the generalization of learned behaviors to new, unseen environments continue to pose considerable hurdles (Jaegle et al., 2021).

3.3 Advanced methods

3.3.1 Adversarial imitation learning

Adversarial Imitation Learning (AIL) offers an efficient and advanced approach in the domain of artificial intelligence, particularly in the context of replicating expert behavior in complex environments. Traditional inverse Reinforcement Learning (IRL) often struggles with the computational complexities that arise when applied to large, complicated environments. These traditional methods are computation-heavy because they typically require solving a reinforcement learning (RL) problem as a subtask, which is both resource-intensive and slow. AIL simplifies this by eliminating the need to solve the RL problem in detail at every iteration, thereby reducing the computational demands significantly (Ho et al., 2016; Finn et al., 2016).

At the heart of AIL is an adversarial framework that orchestrates a dynamic game between two entities: an agent and a discriminator. The discriminator's role is to differentiate between trajectories generated by the agent and those demonstrated by the expert, while the agent strives to produce trajectories indistinguishable from the expert's, thereby incrementally refining its behavior towards an optimal policy that mirrors expert behavior. This adversarial process, inspired by the principles underlying Generative Adversarial Networks (GANs), has been instrumental in the methodological innovations and advancements in AIL (Deka et al., 2023). One of the landmark methodologies in AIL, Generative Adversarial Imitation Learning (GAIL), leverages a discriminator network to differentiate between the behaviors of the agent and the expert, using the confusion of the discriminator as a reward signal to guide the agent toward expert-like behavior. This approach, along with subsequent enhancements, has significantly improved sample efficiency, scalability, and robustness in AIL applications (Fu et al., 2017). Equation 10, highlighted the mentioned approach.

Minimize:
$$L = \sum (action by agent - action by expert)^2$$
 (10)

Where *L* is the loss indicating the difference in actions between the agent and the expert, minimized throughout training.

Recent advancements in AIL have focused on addressing practical challenges such as training instability and reward bias. The introduction of Support-weighted Adversarial Imitation Learning (SAIL) extends traditional AIL algorithms with expert policy support estimation, which enhances the quality of reinforcement signals and improves both performance and training stability across a variety of control tasks (Wang et al., 2020). This improvement potentially applies to agriculture by enhancing robotic precision in tasks such as planting, weeding, and harvesting, thereby increasing efficiency and reducing resource usage. Similarly, the development of Sample-efficient Adversarial Imitation Learning utilizes selfsupervised representation learning to generate robust state and action representations, significantly improving the performance of imitation learning with reduced sample complexity (Jung et al., 2023). In agriculture, this could mean rapidly training models to recognize and act upon diverse crop and soil conditions, even with limited training data, thus enhancing adaptability and responsiveness to environmental changes. Orsini et al. (2021) emphasize the importance of various factors that contribute to the effectiveness of AIL, such as the balance between the generator and discriminator, and the stability of training processes. Their findings suggest that addressing these factors can significantly improve the performance and reliability of AIL methods. Furthermore, the integration of AIL with other learning paradigms, such as hybrid models that leverage the strengths of both AIL and traditional learning approaches, offers promising avenues for addressing broader social challenges and contributing to the global good. This multidisciplinary approach highlights the adaptability and versatility of AIL in various domains, from autonomous driving, where safety and ethical accountability are paramount, to healthcare, where AIL can enhance patient care simulation and treatment planning (Torabi et al., 2018; Yu et al., 2020).

3.3.2 Diffusion policy

As mentioned before, IL in agricultural robotics aims to teach machines to emulate human actions or procedures demonstrated under various operational scenarios. The Diffusion Policy, a novel approach based on conditional denoising diffusion processes, innovatively extends visuomotor policy learning to complex, high-dimensional action spaces typical in agriculture, such as autonomous harvesting or precision farming (Chi et al., 2023). This technique notably surpasses traditional methods by its robust ability to handle multimodal action distributions and to maintain stability during training (Ho et al., 2020).

The Diffusion Policy is grounded in a generative model framework that iteratively refines predictions through a sequence of stochastic corrections, fundamentally based on Stochastic Langevin Dynamics. This process termed the conditional denoising diffusion process, involves an intricate computation of the gradient of the action-distribution score function. Such a calculation not only facilitates the handling of multimodal and high-dimensional action distributions but also enhances the learning efficiency and adaptability required for complex agricultural tasks (Song and Ermon, 2019; Chen, 2023). The denoising process can be mathematically represented in Equation 11 as follows:

$$x_{k-1} = x_k + \eta \nabla_x \log p_k(x_k \mid x_0) + \epsilon_k, \tag{11}$$

Where $\epsilon_k \sim \mathcal{N}(0, \sigma^2 I)$ is the Gaussian noise, η is the step size, and $\nabla_x \log p_k(x_k \mid x_0)$ is the gradient of the log-probability of the action given the initial state, computed at step *k*.

Traditional imitation learning approaches are often limited by the lack of flexibility in handling the diversity of agricultural environments and tasks. The Diffusion Policy addresses these limitations effectively:

- Expressing Multimodal Action Distributions: By leveraging the calculated gradient of the action score function and employing Stochastic Langevin Dynamics for sampling, the Diffusion Policy can proficiently represent and navigate through complex, multimodal action distributions. This feature is crucial for performing varied tasks where the environmental variability demands a high degree of adaptability in action planning (Chen, 2023; Neal, 2011).
- Adaptation to High-dimensional Action Spaces: The inherent capability of diffusion models to scale to high-dimensional spaces allows the policy to infer comprehensive sequences of actions. This scalability is essential for extended operations across large agricultural fields, where tasks may involve complex maneuvering and extended sequences of actions (Dosovitskiy et al., 2020; He et al., 2016).
- Stable Training Regime: Unlike traditional models that often face training instabilities due to the complexities of energybased sampling methods, the Diffusion Policy stabilizes the training process by directly optimizing the gradient of the energy function. This methodological improvement is vital for maintaining consistent performance in the highly variable conditions typical of agricultural settings (Du et al., 2020; Gupta et al., 2019).

Implementing the Diffusion Policy within agricultural robotics involves several technical considerations to optimize its effectiveness in real-world settings:

- Precision Task Execution: The model's high-dimensional action inference capability, combined with its robust training stability, allows for precise and efficient task execution. This precision is particularly beneficial for agriculture tasks that require exact movements, such as selective harvesting and precision pesticide application.
- Autonomous Navigation: Given its proficiency in handling complex distributions and high-dimensional data, the Diffusion Policy is ideally suited for autonomous navigation tasks in agriculture. This includes adapting to varying terrain types, crop densities, and other environmental factors that typically challenge traditional navigation systems.
- Integration with Robotic Systems: The practical application of the Diffusion Policy also involves its integration with existing robotic systems, which may require modifications to the robots' sensory and processing units to fully leverage the capabilities of the policy (Urain et al., 2023; Karras et al., 2022).

The deployment of the Diffusion Policy in agricultural fields brings forth challenges that stem from the need for robust data collection, handling of environmental variability, and integration with diverse robotic technologies. Moreover, the adaptation of this policy to meet the specific demands of agricultural environments requires further empirical studies and technological innovations (Zeng et al., 2021; Zhang et al., 2020). The Diffusion Policy significantly advances imitation learning for agricultural applications, offering solutions that enhance the robustness, versatility, and precision of robotic tasks. This technology holds the potential to revolutionize agricultural practices by improving operational efficiency and reducing labor needs, paving the way for future innovations in robotic farming (Chi et al., 2023; Pearce et al., 2023).

Tables 2, 3 provide a summary of the critical analysis, emphasizing the technical capabilities and limitations of primary and advanced imitation learning models, respectively. These tables highlight how each model performs in the context of dynamic agricultural environments, as discussed in this section.

4 Current applications of IL in agricultural and biological engineering

In the preceding section, we discussed various imitation learning methods and techniques, detailing their potential to significantly enhance agricultural practices. Despite the advantages that imitation learning offers to the field, its application within agriculture remains relatively nascent, with few research endeavors exploring its full potential. This section delves into the most recent research in this area, explores how these innovations can be expanded and suggests new subjects for future investigation.

Recent studies reveal a rising interest in bringing advanced technologies like robotics and AI into agriculture, especially when it comes to using IL to boost efficiency and precision in farming. In the comprehensive review by Starostin et al. (2023), the authors highlight the increasing global focus on developing robotic platforms for agricultural purposes, particularly in countries like the United States, China, and Germany, where research and development in this field have accelerated in recent years. The updated findings from this research are illustrated in Figure 4.

The Figure 5 provided illustrates the global distribution of research publications in agricultural robotics, highlighting the significant interest and investment in this field by various countries. Notably, the Republic of China leads with 43.3% of the total publications, indicating a strong commitment to advancing robotic farming technologies. The United States follows with 21.6%, while Japan, Germany, and Italy also show considerable contributions. These countries are not only at the forefront of research but have also established or are developing policies to support the implementation of these innovations. To ensure the strategic development of agricultural robotics, future policies in key countries are expected to focus on enhancing research in intelligent robotics, fostering innovation in sensor and automation technologies, and supporting the integration of robotics into large-scale agricultural production systems (Zhao et al., 2020).

As we look to the future, with policies aiming to speed up the adoption of advanced technologies in agricultural robotics, it becomes more important than ever to explore new and innovative ways to train these systems effectively. The early application of imitation learning in agriculture, as demonstrated by Dyrstad et al., where robots were instructed to grasp real fish, illustrates the substantial potential of utilizing virtual reality as a sophisticated training tool (Dyrstad et al., 2018). This innovative approach

Aspect	Behavioral cloning (BC)	Inverse reinforcement learning (IRL)	Imitation from observation (IfO)
Scalability	Efficient with structured, large datasets but necessitates diverse data to prevent overfitting (Ross et al., 2011)	Scales effectively in data-rich environments, demanding high computational resources (Arora and Doshi, 2021)	Demonstrates scalability with extensive visual or sensor data (Liu et al., 2018)
Adaptability	Quickly adapts to environments similar to its training data but struggles with novel scenarios (Kim et al., 2023)	Adapts to new tasks by learning underlying objectives, though re-derivation can be cumbersome (Arora and Doshi, 2021)	Shows adaptability in familiar settings but faces limitations with highly divergent new states (Hu et al., 2022)
Performance	Excels in environments resembling the training dataset but limited in dynamic conditions (Pomerleau, 1988)	Offers robust performance across varied scenarios, heavily dependent on precise reward estimation (Shao and Er, 2012)	Achieves high accuracy with well-modeled state transitions but is vulnerable if transitions are not accurately captured (Liu et al., 2018)
Generalization	Generalizes effectively within the scope of its training data but is constrained by limited diversity (Ross et al., 2011)	Excellent potential for generalization, reliant on the universality of inferred rewards (Skalse and Abate, 2023)	Generalizes well from observed to similar unobserved states, limited by the diversity of states during training (Edwards et al., 2019)
Dynamic Handling	Effective in dynamic settings if continuously updated with new data, but requires frequent retraining (Park and Yang, 2022)	Handles dynamic environments by learning applicable rewards but requires complex recalculations (Freymuth et al., 2021)	Capable in dynamic settings if regularly updated with new data, though demands continuous retraining (Liu et al., 2018)

TABLE 2 Critical comparison of primary imitation learning models in agricultural applications, highlighting key aspects of scalability, adaptability, and performance.

TABLE 3 Critical comparison of advanced imitation learning models in agricultural applications, focusing on adaptability, performance, and dynamic handling.

Aspect	Adversarial imitation learning (AIL)	Diffusion policy
Scalability	Initial resource intensity decreases as more diverse data becomes available, enhancing scalability (Ho et al., 2016)	Handles high-dimensional action spaces effectively, though computational demands may constrain scalability (Chi et al., 2023)
Adaptability	Highly adaptable via its adversarial process, yet maintaining balance and training quality is crucial (Finn et al., 2016)	Adapts proficiently to complex action distributions but requires significant retraining for new task types (Chi et al., 2023)
Performance	Can outperform other models by learning nuanced behavior patterns but may experience instability during early training phases (Ho et al., 2016)	Excels in handling multimodal actions; however, its performance consistency heavily relies on the quality of stochastic modeling (Ho et al., 2020)
Generalization	Provides strong generalization across diverse scenarios when adequately trained, though there is a risk of overfitting (Fu et al., 2017)	Proficient at generalizing across high-dimensional and multimodal environments, necessitating thorough training (Song and Ermon, 2019)
Dynamic Handling	Adapts dynamically through continuous adversarial training, necessitating consistent data input and adjustments (Finn et al., 2016)	Manages dynamic changes efficiently through stochastic corrections but requires ongoing calibration (Song and Ermon, 2019)

could be effectively expanded to encompass a range of complex agricultural tasks, particularly with the daily improvement of Machine Learning methods in agricultural product prediction (Khan et al., 2020). These improvements, combined with predictive models for agricultural production, offer valuable tools for policymakers and farmers to enhance fruit production, optimize resource management, and address the growing demand for agricultural efficiency (Rehman et al., 2018). For instance, it could be adapted for the harvesting of irregularly shaped or delicate fruits and vegetables, thereby opening new avenues for research in adaptive robotic handling techniques. Additionally, the implementation of imitation learning holds considerable promise for challenging environments such as poultry plants or crab meat harvesting operations. These fields, characterized by their harsh working conditions, could greatly benefit from the precision and adaptability that imitation learning-equipped robots can offer, potentially transforming labor-intensive processes into more efficient, automated operations.





Kim, Ohmura, and Kuniyoshi (2022a) made significant strides in the robotic handling of delicate tasks with the development of a sophisticated robotic system for peeling bananas. This system employs goal-conditioned dual-action deep imitation learning and features a dual-arm robot setup, where one arm stabilizes the banana while the other expertly peels it. This complex coordination between the arms is crucial for handling tasks that require precision and gentle handling, demonstrating the potential for similar robotic systems to be adapted for processing other agricultural products that necessitate delicate operations, such as peeling or slicing, thus minimizing waste and maximizing efficiency.

Complementing this, Kim J. et al. (2022) introduced the Deep-ToMaToS, a deep learning network tailored for the 6D pose estimation of tomatoes, enabling precise robotic harvesting based on maturity classification. This approach highlights the capability of deep learning models to significantly enhance the accuracy and efficiency of robotic systems in agriculture, especially in recognizing and classifying crops according to their maturity. Together, these studies showcase the potential of advanced machine learning models to revolutionize the automation of intricate agricultural tasks, from the precise handling and processing of individual fruits to the efficient harvesting of crops, opening avenues for these technologies to be applied across a broader spectrum of agricultural needs.

The research conducted by Tsai et al. (2018) in both 2018 and 2019 significantly advanced the precision and adaptability of agricultural robots through the development of visual picking controls using end-to-end imitation learning. In their 2018 study, Tsai et al. (2019) focused on enabling a 6-DoF (Degree of Freedom) manipulator to adapt its picking strategies based on real-time visual cues. This approach allowed the robot to effectively identify and pick objects in a dynamic environment, which is crucial for applications such as fruit or vegetable harvesting where conditions can vary widely. Extending these capabilities, another research introduced an omnidirectional mobile manipulator equipped with enhanced multi-task learning capabilities. This system was not only capable of navigating across different terrains but could also perform various agricultural tasks simultaneously, thereby increasing efficiency and reducing the need for human intervention.

These enhancements in robotic technology suggest that the methods developed by Tsai et al. could be applied to a broader range of agricultural robots, particularly those used for seeding and weeding. By adapting the visual picking controls and multitask learning approaches, these robots could potentially operate autonomously within fully automated farm environments. Such advancements could lead to the creation of multi-robot systems where various agricultural tasks, from planting to pest control, are allocated efficiently among different robots. Each robot could learn from human experts through imitation, thereby continually improving its operational efficiency and adaptability.

Moreover, the foundational work by Tsai et al. opens up opportunities for future research into the collaborative operation of these robots in a coordinated system. This could involve the development of algorithms that enable robots to learn not only from human demonstrations but also from each other, optimizing task allocation and execution in real-time based on environmental conditions and ongoing learning. Such systems would represent a significant step toward the realization of smart agriculture ecosystems, where multiple robotic agents work in concert to manage and optimize various farming operations, leading to higher productivity and sustainability.

Misimi et al. (2018) and Porichis et al. (2023) both explore the capabilities of robotic systems in handling specific agricultural products, emphasizing the importance of delicate handling to maintain product integrity and quality. Misimi et al. (2018) developed robust learning-from-demonstration techniques, initially focused on the handling of fish and other malleable materials. This approach utilizes visual and tactile feedback to adapt to the varied textures and consistencies encountered in these materials, which is crucial for tasks such as filleting or packaging in food production environments. The techniques demonstrated by Misimi et al. (2018) have significant potential for extension to the handling and processing of plants, particularly in applications where the preservation of structural integrity is critical, such as in the handling of soft fruits and vegetables. On the other hand, Porichis et al., 2023 addressed the specific challenges associated with mushroom harvesting by developing a visual imitation learning system that not only recognizes mushrooms but also picks them with precision. Their system is designed to minimize damage to the mushrooms, thereby improving yield and product quality. The advanced visual recognition capabilities and delicate handling methods are particularly important for crops like mushrooms, which are susceptible to bruising and damage during the picking process. Both studies underscore the broader applicability of these robotic technologies to other types of crops that require gentle handling. By adapting the visual and tactile-based imitation learning techniques developed in these studies, it is conceivable to design robotic systems that could handle a wider array of delicate agricultural products, from berries to leafy greens, reducing labor requirements and enhancing product quality. These advancements hold the promise of transforming agricultural practices by introducing more efficient, precise, and gentle robotic systems into the harvesting and processing stages, ultimately leading to more sustainable and productive agricultural operations.

Zhou et al., 2022a reviewed the latest advancements and persistent challenges in the development of robotic systems for fruit harvesting. This paper underscores the critical need for robots that are not only intelligent but also highly adaptive, capable of handling the diverse and often unpredictable conditions typical of agricultural environments. The authors emphasize the complexity of accurately identifying and harvesting different types of crops, which vary widely in size, shape, and the degree of delicacy required during handling. The study encourages further research into integrating multi-sensor fusion techniques and artificial intelligence to improve the autonomy and efficiency of harvesting robots. By doing so, these machines could potentially learn from each harvest, optimizing their paths and techniques based on accumulated experiences and data, thereby reducing waste and increasing yield. This emphasis on the need for greater intelligence and adaptability in robotic harvesting systems calls for a multidisciplinary approach, combining robotics, machine learning, and agricultural sciences to develop solutions that can sustainably meet the demands of global food production.

Given the nascent state of imitation learning in agriculture, as evidenced by the literature, there are ample opportunities for future investigations. These studies could focus on developing more sophisticated models of learning that not only imitate simple actions but also complex decision-making processes, thereby enhancing the cognitive capabilities of agricultural robots. As agricultural demands evolve alongside technological advancements, imitation learning is poised to become a pivotal technology in transforming agricultural practices, addressing some of the most pressing challenges in the field.

5 Challenges and ethical considerations

As agricultural robotics and IL evolve, they present a multitude of challenges and ethical considerations that necessitate thoughtful deliberation and proactive management. This chapter aims to highlight the primary obstacles in the deployment of IL technologies in agriculture, address the ethical implications, and propose directions for future research and development.

5.1 Data acquisition and quality

A major hurdle in implementing IL in agricultural robotics is securing high-quality data (Yang et al., 2021). The process of gathering substantial volumes of annotated data in agricultural settings is notably laborious and time-consuming, demanding considerable human expertise. Moreover, the quality of data can be compromised by human errors due to distractions or limited visibility and comprehension of complex environments (Wu et al., 2019; Sasaki and Yamashina, 2020).

Crowd-sourced datasets, often employed to boost the robustness and efficacy of IL policies, introduce additional challenges due to their inherent variability. These datasets display a wide range of demonstration qualities, mirroring the varying expertise of contributors (Ramrakhya et al., 2022). While discarding lower-quality demonstrations might appear to be a straightforward fix, it is rarely feasible due to the massive manual effort required for data curation, which is typically impractical (Sasaki and Yamashina, 2020).

Furthermore, often the diverse and unstructured nature of agricultural environments-from open fields to greenhouses-poses unique challenges for data collection. The need for adaptable data collection strategies is accentuated by environmental variations such as lighting, weather, and seasonal changes, which can significantly affect the physical characteristics of crops and disrupt data consistency across different periods (Ramrakhya et al., 2022; Bechar and Vigneault, 2016). Adverse weather conditions like rain and fog can also impair the quality and reliability of sensor data. Moreover, the performance of commonly used sensors in agricultural robotics, including cameras, LiDAR, and GPS, may be hindered by limitations in resolution, range, and accuracy under challenging conditions such as dust, rain, and fluctuating light levels (Ozdogan et al., 2010; Bargoti and Underwood, 2017). These issues are critical as they directly influence the quality of data essential for training dependable IL models.

5.2 Generalization and adaptability

Ensuring IL models generalize effectively across diverse agricultural settings presents a significant challenge. Models trained

in specific environments may not perform effectively when deployed in new or different settings due to variations in crop types, growth stages, and farming practices. This issue, often referred to as the "domain shift" problem, necessitates the development of models that can adapt in real time to dynamic and unpredictable agricultural environments.

Addressing domain discrepancies is a focal point of recent IL research, with studies exploring dynamics, viewpoint, and embodiment mismatches Kim et al. (2020); Stadie et al. (2017). These efforts typically involve developing mappings between stateaction spaces to foster domain-invariant features, which are crucial for successful policy learning across varying conditions. The need for feature-level domain adaptation is discussed by Kouw et al. (2016), who propose an approach that models the dependence between two domains through a feature-level transfer model, aiming to reduce the feature distribution discrepancies that are common in diverse agricultural settings. Moreover, Nguyen et al. (2021) introduced TIDOT, a model that employs the principles of imitation learning and optimal transport for unsupervised domain adaptation. This model demonstrates how a "teacher" agent in the source domain can guide a "student" agent in the target domain, thereby enabling realtime adaptation to new agricultural environments. These advanced methodologies, including Cross-embodiment IRL (XIRL) and other approaches by Fickinger et al. (2021), focus on extracting taskspecific policies that are resilient to variations in environment dynamics and embodiment (Beliaev et al., 2022; Kim et al., 2020).

5.3 Computational constraints

Addressing the computational constraints in deploying IL models on agricultural robots is crucial due to the limited computational resources and energy availability inherent to field robots. Balancing the performance of high-complexity models with the constrained computational power and battery life of agricultural robots is essential for practical applications.

To tackle these challenges, researchers have focused on model compression and optimization techniques that maintain model performance while reducing computational demands and energy consumption. Techniques such as quantization, pruning, and knowledge distillation are particularly effective in adapting IL models to the constraints of agricultural robots (Tokekar et al., 2016). These methods aim to decrease the model size and computational requirements, thus making IL models more suitable for deployment on resource-constrained robots in the field.

Liu et al. (2021) discussed the application of pruning and quantization to reduce the computational load of models without significantly sacrificing performance, making them feasible for implementation on robots with limited processing capabilities and battery life. Similarly, Tokekar et al. (2016) explored the balance between model complexity and computational feasibility, emphasizing the importance of efficient energy management in prolonging the operational time of field robots. Further research by Buchli et al. (2017) and McCool et al. (2016) has highlighted the critical role of energy-efficient model design in ensuring that robots can perform extended tasks without frequent recharging. These studies suggest that advanced model optimization strategies, beyond traditional compression techniques, are necessary to achieve the dual goals of high performance and low energy consumption in agricultural settings.

Recent developments in this area have also included more sophisticated approaches such as differentiable constrained IL for robot motion planning and control, which integrate hard constraints into the learning process to ensure safe and efficient robot operation under limited computational resources (Diehl et al., 2022).

5.4 Ethical and social considerations

The deployment of agricultural robots brings forth profound ethical and social considerations, including issues of labor displacement, data privacy, and the broader implications on societal structures. The integration of these technologies can significantly affect employment, amplify existing social inequalities through algorithmic biases, and challenge the governance frameworks needed to ensure safe and equitable use.

The discussion around the moral status of robots, as introduced by Bostrom and Yudkowsky (2018), emphasizes the potential for robots to alter societal norms and structures dramatically. They suggest that without careful consideration, the deployment of autonomous systems could exacerbate social inequalities and undermine human dignity by making biased decisions or behaving unpredictably in human environments. This necessitates robust ethical frameworks to ensure that robotics technologies are aligned with human values and safety standards.

As the agricultural sector transitions from labor-intensive methods to automation, the role of robots has expanded. This shift, discussed by Marinoudi et al. (2019), raises significant questions about the effects on employment and the social fabric of rural communities. As repetitive tasks such as fruit picking and weeding are increasingly automated, many workers may find their roles replaced by machines. Reskilling programs are urgently needed to help these workers transition to new roles, such as robot maintenance and operation. De Stefano (2019) suggests that without these interventions, rural communities may experience increased income disparities and social disruption. The introduction of automation can also widen the gap between large, technology-driven farms and smaller farms that lack the resources to adopt these innovations, potentially leading to socioeconomic inequality (Dauth et al., 2021).

In many cases also, agricultural roles may be augmented rather than entirely replaced, with robots starting to work alongside humans. This collaborative approach could mitigate some negative impacts on employment but requires careful management and supportive policies to ensure equitable outcomes. New job opportunties will also be created in technology-driven roles such as robot maintenance, programming, and system management. Dauth et al. (2021) observed that while robots have displaced certain jobs in manufacturing, they have also generated new, often higher-quality positions in the service sector. This suggests that educational and training programs are crucial in preparing displaced workers for new opportunities in an increasingly automated agricultural landscape.

The digitization of agriculture also brings to the forefront issues surrounding the privacy and security of farm data. As farming practices and crop management become more data-driven,

the risk of sensitive information falling into unauthorized hands increases. Strong data security, anonymization protocols, and clear data governance structures are essential to protect farmers' privacy and ensure that the collected information is used responsibly. Data ownership is a critical issue, especially as farmers increasingly rely on third-party platforms to analyze their data. When farmers outsource data processing to these platforms, they risk losing control over their data, potentially exposing them to exploitation or misuse. Rose et al. (2021) emphasize the importance of embedding ethical considerations early in the design and development phase of autonomous agricultural robots to safeguard data integrity and ensure farmer privacy. Strong data security measures, such as encryption and anonymization, are essential to protect farmers' information. However, data ownership is a growing issue, especially as farmers rely on third-party platforms to analyze and process their data. When this information is uploaded to external systems, farmers risk losing control over it. Kshetri (2014) points out that data ownership policies need to be clearly defined, with specific regulations governing who has access and rights to the data collected by agricultural robots. Clear legal frameworks should ensure that farmers retain control over their data and that it cannot be misused by corporations or third-party platforms (Wolfert et al., 2017).

The societal impact of robotics in agriculture also extends beyond employment. Klerkx and Rose (2020) argue that automation may lead to land consolidation, as smaller farms struggle to compete and are forced to sell their land to larger agribusinesses. This shift could reduce the autonomy of smallholder farmers and weaken the social fabric of rural communities. It is essential that policymakers ensure equitable access to robotics and AI technologies to prevent this divide.

Addressing the ethical and social implications of agricultural robots requires effective policymaking that aims to distribute the benefits of automation broadly across society, supporting both displaced workers and the communities impacted by these technological changes. Sparrow and Howard (2020) emphasize the need for key policy choices that maximize the social, environmental, and economic benefits of agricultural robotics while mitigating potential harms. The integration of robotics in agriculture thus presents a complex landscape of ethical and social challenges, including potential labor displacement, the creation of new technological roles, concerns over data privacy, and the need for robust governance frameworks. A comprehensive and balanced approach is required to ensure that the advancement of robotics in agriculture aligns with human values and promotes equitable, socially just outcomes. This approach will be critical in navigating the evolving moral landscape as robotics become more embedded in daily farming operations.

6 Conclusion and future directions

This survey on imitation learning in agricultural robotics encapsulates the transformative potential and innovative impact of this technology in revolutionizing agricultural practices. IL adeptly mimics expert human strategies in complex and variable environments, enhancing efficiency, adaptability, and productivity in agricultural operations. Techniques such as AIL and Diffusion Policy in IL are significant advancements, in improving the practical deployment of robotic systems in farming.

Despite these advancements, the application of IL in agriculture faces substantial challenges including data collection in diverse environments, the need for model generalization across different agricultural contexts, and computational constraints limiting field deployment. Moreover, the ethical implications surrounding labor displacement and data privacy necessitate a balanced approach to technology integration.

Future research should address these challenges by enhancing model adaptability, improving data collection quality through advanced sensing technologies, and optimizing IL models for computational efficiency using techniques such as model compression and quantization. Additionally, developing comprehensive ethical frameworks and policy guidelines is crucial to address socio-economic impacts and ensure equitable benefits from agricultural robotics.

Exploring collaborative human-robot interactions and expanding IL applications to include a broader range of agricultural tasks, such as pest management and environmental monitoring, will also be pivotal. These efforts will not only mitigate job displacement concerns but will also enhance the acceptance and effectiveness of robotic technologies in agriculture.

By focusing on these areas, the future of IL in agricultural robotics can achieve its potential, leading to revolutionary advancements in agricultural practices. This will contribute to more sustainable, productive, and less labor-intensive farming methods globally, aligning technological progress with human values and environmental needs.

Author contributions

SM: Conceptualization, Formal Analysis, Methodology, Writing-original draft, Writing-review and editing. AD:

References

Alshawabkeh, M., Alagi, H., Navarro, S. E., Duriez, C., Hein, B., Zangl, H., et al. (2023). Highly stretchable additively manufactured capacitive proximity and tactile sensors for soft robotic systems. *IEEE Trans. Instrum. Meas.* 72, 1–10. doi:10.1109/tim.2023.3250232

Arad, B., Balendonck, J., Barth, R., Ben-Shahar, O., Edan, Y., Hellström, T., et al. (2020). Development of a sweet pepper harvesting robot. *J. Field Robotics* 37, 1027–1039. doi:10.1002/rob.21937

Argall, B. D., and Billard, A. G. (2010). A survey of tactile human-robot interactions. *Robotics Aut. Syst.* 58, 1159–1176. doi:10.1016/j.robot.2010.07.002

Arora, S., and Doshi, P. (2021). A survey of inverse reinforcement learning: challenges, methods and progress. *Artif. Intell.* 297, 103500. doi:10.1016/j.artint.2021.103500

Aytar, Y., Pfaff, T., Budden, D., Paine, T., Wang, Z., and De Freitas, N. (2018). Playing hard exploration games by watching youtube. *Adv. neural Inf. Process. Syst.* 31. doi:10.48550/arXiv.1805.11592

Baldini, G., Staiano, M., Grella, F., Frascio, M., Maiolino, P., and Cannata, G. (2023). Mathematical model and experimental characterization of vertically stacked capacitive tactile sensors. *IEEE Sensors J.* 23, 21341–21354. doi:10.1109/jsen.2023.3300363

Balesni, M., Tampuu, A., and Matiisen, T. (2023). Controlling steering with energybased models. *arXiv Prepr.* doi:10.48550/arXiv.2301.12264

Bansal, M., Krizhevsky, A., and Ogale, A. (2018). Chauffeured: learning to drive by imitating the best and synthesizing the worst. *arXiv preprint arXiv:1812.03079*

Bargoti, S., and Underwood, J. (2017). "Deep fruit detection in orchards," in 2017 IEEE international conference on robotics and automation (ICRA) (IEEE), 3626–3633. Visualization, Writing-review and editing. PS: Writing-review and editing. RB: Writing-review and editing. YT: Funding acquisition, Writing-review and editing. DW: Funding acquisition, Project administration, Resources, Supervision, Writing-review and editing.

Funding

The author(s) declare that financial support was received for the research, authorship, and/or publication of this article. This research was supported by the intramural research program of the U.S. Department of Agriculture, National Institute of Food and Agriculture, Agriculture and Food Research Initiative 2023-70442-39232 and 2023-67021-40621.

Conflict of interest

The authors declare that the research was conducted in the absence of any commercial or financial relationships that could be construed as a potential conflict of interest.

Publisher's note

All claims expressed in this article are solely those of the authors and do not necessarily represent those of their affiliated organizations, or those of the publisher, the editors and the reviewers. Any product that may be evaluated in this article, or claim that may be made by its manufacturer, is not guaranteed or endorsed by the publisher.

Bayer, I. S. (2022). Mems-based tactile sensors: materials, processes and applications in robotics. *Micromachines* 13, 2051. doi:10.3390/mi13122051

Bechar, A., and Vigneault, C. (2016). Agricultural robots for field operations: concepts and components. *Biosyst. Eng.* 149, 94–111. doi:10.1016/j.biosystemseng.2016.06.014

Beliaev, M., Shih, A., Ermon, S., Sadigh, D., and Pedarsani, R. (2022). "Imitation learning by estimating expertise of demonstrators," in *International conference on machine learning* (MD, USA: PMLR, July 17–23, 2022), 1732–1748. doi:10.5555/3635637.3663028

Bostrom, N., and Yudkowsky, E. (2018). "The ethics of artificial intelligence," in *Artificial intelligence safety and security* (Chapman and Hall/CRC), 57–69.

Botta, A., Cavallone, P., Baglieri, L., Colucci, G., Tagliavini, L., and Quaglia, G. (2022). A review of robots, perception, and tasks in precision agriculture. *Appl. Mech.* 3, 830–854. doi:10.3390/applmech3030049

Brewster, C., Roussaki, I., Kalatzis, N., Doolin, K., and Ellis, K. (2017). Iot in agriculture: designing a europe-wide large-scale pilot. *IEEE Commun. Mag.* 55, 26–33. doi:10.1109/mcom.2017.1600528

Broussard, M. A., Coates, M., and Martinsen, P. (2023). Artificial pollination technologies: a review. *Agronomy* 13, 1351. doi:10.3390/agronomy13051351

Buchli, J., Farshidian, F., Winkler, A., Sandy, T., and Giftthaler, M. (2017). Optimal and learning control for autonomous robots. *arXiv preprint arXiv:1708.09342*

Chang, J., Uehara, M., Sreenivas, D., Kidambi, R., and Sun, W. (2021). Mitigating covariate shift in imitation learning via offline data with partial coverage. *Adv. Neural Inf. Process. Syst.* 34, 965–979. doi:10.48550/arXiv.2106.03207

Chen, T. (2023). On the importance of noise scheduling for diffusion models. arXiv preprint arXiv:2301.10972

Cheng, C., Fu, J., Su, H., and Ren, L. (2023). Recent advancements in agriculture robots: benefits and challenges. *Machines* 11, 48. doi:10.3390/machines11010048

Chi, C., Feng, S., Du, Y., Xu, Z., Cousineau, E., Burchfiel, B., et al. (2023). Diffusion policy: visuomotor policy learning via action diffusion. *arXiv preprint arXiv:2303.04137*

Choe, S., Seong, H., and Kim, E. (2021). Indoor place category recognition for a cleaning robot by fusing a probabilistic approach and deep learning. *IEEE Trans. Cybern.* 52, 7265–7276. doi:10.1109/tcyb.2021.3052499

Daaboul, K., Becker, S., Daiss, K., Kurzer, K., Geimer, M., and Zöllner, J. M. (2019). Machine learning for process automation of agricultural machines in field applications

Dadashi, R., Hussenot, L., Geist, M., and Pietquin, O. (2020). Primal wasserstein imitation learning. *arXiv Prepr. arXiv:2006*. doi:10.48550/arXiv.2006.04678

Dauth, W., Findeisen, S., Suedekum, J., and Woessner, N. (2021). The adjustment of labor markets to robots. *J. Eur. Econ. Assoc.* 19, 3104–3153. doi:10.1093/jeea/jvab012

Deka, A., Liu, C., and Sycara, K. P. (2023). "Arc-actor residual critic for adversarial imitation learning," in *Conference on robot learning (PMLR)*, 1446–1456.

DeMoss, B., Duckworth, P., Hawes, N., and Posner, I. (2023). Ditto: offline imitation learning with world models. *arXiv preprint arXiv:2302.03086*

De Stefano, V. (2019). "Negotiating the algorithm": automation, artificial intelligence, and labor protection. *Comp. Lab. L. and Poly J.*, 41, 15. doi:10.2139/ssrn.3178233

Dexter, G., Bello, K., and Honorio, J. (2021). Inverse reinforcement learning in a continuous state space with formal guarantees. *Adv. Neural Inf. Process. Syst.* 34, 6972–6982. doi:10.48550/arXiv.2102.07937

Diehl, C., Adamek, J., Krüger, M., Hoffmann, F., and Bertram, T. (2022). Differentiable constrained imitation learning for robot motion planning and control. *arXiv Prepr. arXiv:2210.* doi:10.48550/arXiv.2210.11796

Dosovitskiy, A., Beyer, L., Kolesnikov, A., Weissenborn, D., Zhai, X., Unterthiner, T., et al. (2020). An image is worth 16x16 words: transformers for image recognition at scale. *arXiv preprint arXiv:2010.11929*

Du, Y., Li, S., Tenenbaum, J., and Mordatch, I. (2020). Improved contrastive divergence training of energy based models. *arXiv preprint arXiv:2012.01316*

Dyrstad, J. S., Øye, E. R., Stahl, A., and Mathiassen, J. R. (2018). in *IEEE/RSJ* international conference on intelligent robots and systems (IROS) (IEEE), 7185–7192. Teaching a robot to grasp real fish by imitation learning from a human supervisor in virtual reality

Edwards, A., Sahni, H., Schroecker, Y., and Isbell, C. (2019). "Imitating latent policies from observation," in *International conference on machine learning* (CA, USA: PMLR, June 9-15, 2019), 1755–1763.

Eizicovits, D., and Berman, S. (2014). Efficient sensory-grounded grasp pose quality mapping for gripper design and online grasp planning. *Robotics Aut. Syst.* 62, 1208–1219. doi:10.1016/j.robot.2014.03.011

Fang, B., Jia, S., Guo, D., Xu, M., Wen, S., and Sun, F. (2019). Survey of imitation learning for robotic manipulation. *Int. J. Intelligent Robotics Appl.* 4, 362–369. doi:10.1007/s41315-019-00103-5

Fickinger, A., Cohen, S., Russell, S., and Amos, B. (2021). Cross-domain imitation learning via optimal transport. *arXiv preprint arXiv:2110.03684*

Finn, C., Christiano, P., Abbeel, P., and Levine, S. (2016). A connection between generative adversarial networks, inverse reinforcement learning, and energy-based models. *arXiv Prepr. arXiv:1611.03852*. doi:10.48550/arXiv.1611.03852

Finn, C., Yu, T., Zhang, T., Abbeel, P., and Levine, S. (2017). "One-shot visual imitation learning via meta-learning," in *Conference on robot learning (PMLR)*, 357–368. doi:10.48550/arXiv.1709.04905

Florence, P., Lynch, C., Zeng, A., Ramirez, O. A., Wahid, A., Downs, L., et al. (2022). "Implicit behavioral cloning," in *Conference on robot learning (PMLR)*, 158–168.

Fountas, S., Mylonas, N., Malounas, I., Rodias, E., Hellmann Santos, C., and Pekkeriet, E. (2020). Agricultural robotics for field operations. *Sensors* 20, 2672. doi:10.3390/s20092672

Freymuth, N., Becker, P., and Neumann, G. (2021). Versatile inverse reinforcement learning via cumulative rewards. *arXiv Prepr. arXiv:2111.07667*. doi:10.48550/arXiv.2111.07667

Fu, J., Luo, K., and Levine, S. (2017). Learning robust rewards with adversarial inverse reinforcement learning. *arXiv preprint arXiv:1710.11248*

Fu, J., Tacchetti, A., Perolat, J., and Bachrach, Y. (2021). Evaluating strategic structures in multi-agent inverse reinforcement learning. *J. Artif. Intell. Res.* 71, 925–951. doi:10.1613/jair.1.12594

Goel, S., Markanday, S., and Mohanty, S. (2022). "Classification of agriculture crops using transfer learning," in 2022 OITS international Conference on information technology (OCIT) (IEEE), 268–272.

Gonzalez-de Santos, P., Fernández, R., Sepúlveda, D., Navas, E., Emmi, L., and Armada, M. (2020). Field robots for intelligent farms—inhering features from industry. *Agronomy* 10, 1638. doi:10.3390/agronomy10111638

Grant, E., Finn, C., Levine, S., Darrell, T., and Griffiths, T. (2018). Recasting gradient-based meta-learning as hierarchical bayes. *arXiv Prepr. arXiv:1801.08930*. doi:10.48550/arXiv.1801.08930

Grieve, B. D., Duckett, T., Collison, M., Boyd, L., West, J., Yin, H., et al. (2019). The challenges posed by global broadaxe crops in delivering smart agrirobotic solutions: a fundamental rethink is required. *Glob. Food Secur.* 23, 116–124. doi:10.1016/j.gfs.2019.04.011

Gupta, A., Kumar, V., Lynch, C., Levine, S., and Hausman, K. (2019). Relay policy learning: solving long-horizon tasks via imitation and reinforcement learning. *arXiv Prepr. arXiv*:1910.11956.

He, K., Zhang, X., Ren, S., and Sun, J. (2016). "Deep residual learning for image recognition," in *Proceedings of the IEEE conference on computer vision and pattern recognition*, 770–778.

Ho, J., and Ermon, S. (2016). Generative adversarial imitation learning. Adv. neural Inf. Process. Syst. 29. doi:10.48550/arXiv.2410.02231

Ho, J., Gupta, J., and Ermon, S. (2016). "Model-free imitation learning with policy optimization," in *International conference on machine learning* (NY, USA: PMLR, June 20-22, 2016), 2760–2769. doi:10.48550/arXiv.1605.08478

Ho, J., Jain, A., and Abbeel, P. (2020). Denoising diffusion probabilistic models. *Adv. neural Inf. Process. Syst.* 33, 6840–6851. doi:10.48550/arXiv.2006.11239

Hoque, R., Balakrishna, A., Novoseller, E., Wilcox, A., Brown, D. S., and Goldberg, K. (2021). Thriftydagger: budget-aware novelty and risk gating for interactive imitation learning. *arXiv Prepr. arXiv:2109.08273*. doi:10.48550/arXiv.2109.08273

Hu, Y., Chen, G., Li, Z., and Knoll, A. (2022). Robot policy improvement with natural evolution strategies for stable nonlinear dynamical system. *IEEE Trans. Cybern.* 53, 4002–4014. doi:10.1109/tcyb.2022.3192049

Hussein, A., Gaber, M. M., Elyan, E., and Jayne, C. (2017). Imitation learning: a survey of learning methods. ACM Comput. Surv. (CSUR) 50, 1–35. doi:10.1145/3054912

Jaegle, A., Sulsky, Y., Ahuja, A., Bruce, J., Fergus, R., and Wayne, G. (2021). "Imitation by predicting observations," in *International conference on machine learning* (PMLR), 4665–4676.

Jung, D., Lee, H., and Yoon, S. (2023). Sample-efficient adversarial imitation learning. arXiv Prepr. arXiv:2303.07846. doi:10.48550/arXiv.2303.07846

Karras, T., Aittala, M., Aila, T., and Laine, S. (2022). Elucidating the design space of diffusion-based generative models. *Adv. Neural Inf. Process. Syst.* 35, 26565–26577. doi:10.48550/arXiv.2206.00364

Kaur, S., and Bawa, G. (2022). "Learning robotic skills through reinforcement learning," in 2022 3rd international conference on electronics and sustainable communication systems (ICESC) (IEEE), 903–908.

Kelly, M., Sidrane, C., Driggs-Campbell, K., and Kochenderfer, M. J. (2019). "Hg-dagger: interactive imitation learning with human experts," in 2019 international conference on robotics and automation (ICRA) (IEEE), 8077–8083.

Khan, T., Qiu, J., Qureshi, M. A. A., Iqbal, M. S., Mehmood, R., and Hussain, W. (2020). Agricultural fruit prediction using deep neural networks. *Procedia Comput. Sci.* 174, 72–78. doi:10.1016/j.procs.2020.06.058

Kim, D., Lee, S., Hong, T. H., and Park, Y.-L. (2023). Exploration-based model learning with self-attention for risk-sensitive robot control. *npj Robot.* 1 (7), 7. doi:10.1038/s44182-023-00006-5

Kim, H., Ohmura, Y., and Kuniyoshi, Y. (2022a). Robot peels banana with goalconditioned dual-action deep imitation learning. *arXiv Prepr. arXiv:2203.09749*. doi:10.48550/arXiv.2203.09749

Kim, J., Pyo, H., Jang, I., Kang, J., Ju, B., and Ko, K. (2022b). Tomato harvesting robotic system based on deep-tomatoes: deep learning network using transformation loss for 6d pose estimation of maturity classified tomatoes with side-stem. *Comput. Electron. Agric.* 201, 107300. doi:10.1016/j.compag.2022.107300

Kim, K., Gu, Y., Song, J., Zhao, S., and Ermon, S. (2020). "Domain adaptive imitation learning," in *International conference on machine learning* (PMLR, July 13-18, 2020), 5286–5295.

Klerkx, L., and Rose, D. (2020). Dealing with the game-changing technologies of agriculture 4.0: how do we manage diversity and responsibility in food system transition pathways? *Glob. Food Secur.* 24, 100347. doi:10.1016/j.gfs.2019.100347

Korsunsky, P., Belogolovsky, S., Zahavy, T., Tessler, C., and Mannor, S. (2019). Inverse reinforcement learning in contextual mdps. *ArXiv* abs/1905.09710

Kouw, W. M., Van Der Maaten, L. J., Krijthe, J. H., and Loog, M. (2016). Feature-level domain adaptation. J. Mach. Learn. Res. 17, 1–32. doi:10.48550/arXiv.1512.04829

Krithiga, M. D., Viswanandhne, S., and Nandhini, P. (2017). A machine learning perspective for predicting agricultural droughts. *Artif. Intelligent Syst. Mach. Learn.* 9, 31–38. https://www.ciitresearch.org/dl/index.php/aiml/article/view/AIML022017003

Kshetri, N. (2014). Big data' s impact on privacy, security and consumer welfare. *Telecommun. Policy* 38, 1134–1145. doi:10.1016/j.telpol.2014.10.002

Le Mero, L., Yi, D., Dianati, M., and Mouzakitis, A. (2022). A survey on imitation learning techniques for end-to-end autonomous vehicles. *IEEE Trans. Intelligent Transp. Syst.* 23, 14128–14147. doi:10.1109/tits.2022.3144867 Li, Q., Peng, Z., and Zhou, B. (2022). Efficient learning of safe driving policy via human-ai copilot optimization. *arXiv Prepr. arXiv:2202.10341*. doi:10.48550/arXiv.2202.10341

Li, X., Laguë, C., and Uchida, T. K. (2023). "Integration of model predictive control and proportional-integral-derivative strategies for the autonomous path tracking of agricultural wide-span implement carriers," in 2023 IEEE international symposium on robotic and sensors environments ROSE (IEEE), 1–7.

Liang, T., Glossner, J., Wang, L., Shi, S., and Zhang, X. (2021). Pruning and quantization for deep neural network acceleration: a survey. *Neurocomputing* 461, 370–403. doi:10.1016/j.neucom.2021.07.045

Lindner, D., Krause, A., and Ramponi, G. (2022). Active exploration for inverse reinforcement learning. ArXiv abs/2207.08645. doi:10.48550/arXiv.2207.08645

Liu, M., Zhao, H., Yang, Z., Shen, J., Zhang, W., Zhao, L., et al. (2021). Curriculum offline imitating learning. *Adv. Neural Inf. Process. Syst.* 34, 6266–6277. doi:10.48550/arXiv.1707.03374

Liu, Y., Gupta, A., Abbeel, P., and Levine, S. (2018). "Imitation from observation: learning to imitate behaviors from raw video via context translation," in 2018 IEEE international conference on robotics and automation (ICRA) (IEEE), 1118–1125.

Malik, D., Palaniappan, M., Fisac, J., Hadfield-Menell, D., Russell, S. J., and Dragan, A. (2018). An efficient, generalized bellman update for cooperative inverse reinforcement learning. In International Conference on Machine Learning. PMLR, 3394–3402.

Mandil, W., Rajendran, V., Nazari, K., and Ghalamzan-Esfahani, A. (2023). Tactilesensing technologies: trends, challenges and outlook in agri-food manipulation. *Sensors* 23, 7362. doi:10.3390/s23177362

Marinoudi, V., Sørensen, C. G., Pearson, S., and Bochtis, D. (2019). Robotics and labour in agriculture. a context consideration. *Biosyst. Eng.* 184, 111–121. doi:10.1016/j.biosystemseng.2019.06.013

McCool, C., Sa, I., Dayoub, F., Lehnert, C., Perez, T., and Upcroft, B. (2016). "Visual detection of occluded crop: for automated harvesting," in 2016 IEEE international conference on robotics and automation ICRA (IEEE), 2506–2512.

Misimi, E., Olofsson, A., Eilertsen, A., Øye, E. R., and Mathiassen, J. R. (2018). "Robotic handling of compliant food objects by robust learning from demonstration," in 2018 IEEE/RSJ international conference on intelligent robots and systems IROS (IEEE), 6972–6979.

Mitka, E. (2018). Strategy for safer agricultural robots. Econ. World 6, 472-481. doi:10.17265/2328-7144/2018.06.006

Moiz, M., Akmal, M., Ishtiaq, M. S., and Javed, U. (2022). "Classification of rice leaves diseases by deep conn-transfer learning approach for improved rice agriculture," in 2022 international conference on emerging trends in electrical, control, and telecommunication engineering ETECTE (IEEE), 1–6.

Neal, R. M. (2011). Micmac using Hamiltonian dynamics. *Handb. markov chain monte carlo* 2, 2.

Ng, A. Y., and Russell, S. (2000). Algorithms for inverse reinforcement learning. *Icml* 1 (2). https://ai.stanford.edu/ ang/papers/icml00-irl.pdf

Nguyen, T., Le, T., Dam, N., Tran, Q. H., Nguyen, T., and Phung, D. Q. (2021). Tidot: a teacher imitation learning approach for domain adaptation with optimal transport. *IJCAI*, 2862–2868. doi:10.24963/ijcai.2021/394

Ohi, N., Lassak, K., Watson, R., Strader, J., Du, Y., Yang, C., et al. (2018). "Design of an autonomous precision pollination robot," in 2018 IEEE/RSJ international conference on intelligent robots and systems (IROS) (IEEE), 7711–7718.

Orsini, M., Raichuk, A., Hussenot, L., Vincent, D., Dadashi, R., Girgin, S., et al. (2021). What matters for adversarial imitation learning? *Adv. Neural Inf. Process. Syst.* 34, 14656–14668. doi:10.48550/arXiv.2106.00672

Ozdogan, M., Yang, Y., Allez, G., and Cervantes, C. (2010). Remote sensing of irrigated agriculture: opportunities and challenges. *Remote Sens.* 2, 2274–2304. doi:10.3390/rs2092274

Page, M. J., McKenzie, J. E., Bossuyt, P. M., Boutron, I., Hoffmann, T. C., Mulrow, C. D., et al. (2021). The prisma 2020 statement: an updated guideline for reporting systematic reviews. *bmj* 372, n71. doi:10.1136/bmj.n71

Park, J., and Yang, L. (2022). Model-based offline imitation learning with non-expert data. *arXiv preprint arXiv:2206.05521*

Pearce, T., Rashid, T., Kanervisto, A., Bignell, D., Sun, M., Georgescu, R., et al. (2023). Imitating human behaviour with diffusion models. *arXiv Prepr. arXiv:2301.* doi:10.48550/arXiv.2301.10677

Pomerleau, D. A. (1988). Alvin: an autonomous land vehicle in a neural network. *Adv. neural Inf. Process. Syst.* 1. https://proceedings.neurips. cc/paper/1988/file/812b4ba287f5ee0bc9d43bbf5bbe87fb-Paper.pdf

Porichis, A., Vasios, K., Iglezou, M., Mohan, V., and Chatzakos, P. (2023). "Visual imitation learning for robotic fresh mushroom harvesting," in 2023 31st mediterranean conference on control and automation (MED) (IEEE), 535–540.

Prakash, A. J., and Prakasam, P. (2023). An intelligent fruits classification in precision agriculture using bilinear pooling convolutional neural networks. *Vis. Comput.* 39, 1765–1781. doi:10.1007/s00371-022-02443-z

Ramrakhya, R., Undersander, E., Batra, D., and Das, A. (2022). "Habitat-web: learning embodied object-search strategies from human demonstrations at scale," in *Proceedings of the IEEE/CVF conference on computer vision and pattern recognition*, 5173–5183.

Raychaudhuri, D. S., Paul, S., Vanbaar, J., and Roy-Chowdhury, A. K. (2021). "Crossdomain imitation from observations," in *International conference on machine learning* (PMLR, July 18-24, 2021), 8902–8912.

Reddy, S., Dragan, A. D., and Levine, S. (2019). Sil: imitation learning via reinforcement learning with sparse rewards. *arXiv Prepr. arXiv:1905.* doi:10.48550/arXiv.1905.11108

Rehman, A., Deyuan, Z., Hussain, I., Iqbal, M. S., Yang, Y., and Jingdong, L. (2018). Prediction of major agricultural fruits production in Pakistan by using an econometric analysis and machine learning technique. *Int. J. Fruit Sci.* 18, 445–461. doi:10.1080/15538362.2018.1485536

Rezk, N. G., Attia, A.-F., El-Rashidy, M. A., El-Sayed, A., and Hemdan, E. E.-D. (2022). An efficient plant disease recognition system using hybrid convolutional neural networks (conns) and conditional random fields (crfs) for smart iot applications in agriculture. *Int. J. Comput. Intell. Syst.* 15, 65. doi:10.1007/s44196-022-00129-x

Robert, M., Thomas, A., and Bergez, J.-E. (2016). Processes of adaptation in farm decision-making models. a review. *Agron. Sustain. Dev.* 36, 64–15. doi:10.1007/s13593-016-0402-x

Rose, D. C., Lyon, J., de Boon, A., Hanheide, M., and Pearson, S. (2021). Responsible development of autonomous robotics in agriculture. *Nat. Food* 2, 306–309. doi:10.1038/s43016-021-00287-9

Ross, S., and Bagnell, J. A. (2014). Reinforcement and imitation learning via interactive no-regret learning. arXiv Prepr. arXiv:1406. doi:10.48550/arXiv.1406.5979

Ross, S., Gordon, G., and Bagnell, D. (2011). "A reduction of imitation learning and structured prediction to no-regret online learning," in *Proceedings of the fourteenth international conference on artificial intelligence and statistics* (FL, USA: JMLR Workshop and Conference Proceedings, April 11-13, 2011), 627–635.

Sasaki, F., and Yamashina, R. (2020). "Behavioral cloning from noisy demonstrations," in *International conference on learning representations*.

Schaal, S. (1999). Is imitation learning the route to humanoid robots? *Trends cognitive Sci.* 3, 233–242. doi:10.1016/s1364-6613(99)01327-3

Sermanet, P., Lynch, C., Chebotar, Y., Hsu, J., Jang, E., Schaal, S., et al. (2018). "Timecontrastive networks: self-supervised learning from video," in 2018 IEEE international conference on robotics and automation (ICRA) (IEEE), 1134–1141.

Shao, Z., and Er, M. (2012). A survey of inverse reinforcement learning techniques. Int. J. Intell. Comput. Cybern. 5, 293–311. doi:10.1108/17563781211255862

Singh, S., Tu, S., and Sindhwani, V. (2023). Revisiting energy based models as policies: ranking noise contrastive estimation and interpolating energy models. *arXiv preprint arXiv:2309.05803*

Skalse, J. M. V., and Abate, A. (2023). Characterising partial identifiability in inverse reinforcement learning for agents with non-exponential discounting

Song, Y., and Ermon, S. (2019). Generative modeling by estimating gradients of the data distribution. *Adv. neural Inf. Process. Syst.* 32. doi:10.48550/arXiv.1907.05600

Sparrow, R., and Howard, M. (2020). Make way for the wealthy? autonomous vehicles, markets in mobility, and social justice. *Mobilities* 15, 514–526. doi:10.1080/17450101.2020.1739832

Spykman, O., Emberger-Klein, A., Gabriel, A., and Gandorfer, M. (2021). Society's view on autonomous agriculture: does digitalization lead to alienation? *Eng. Proc.* 9, 12.

Stadie, B. C., Abbeel, P., and Sutskever, I. (2017). Third-person imitation learning. arXiv Prepr. arXiv:1703.01703. doi:10.48550/arXiv.1703.01703

Starostin, I., Eshchin, A., and Davydova, S. (2023). Global trends in the development of agricultural robotics. In, *IOP Conf. Ser. Earth Environ. Sci.*, vol. 1138, 012042, doi:10.1088/1755-1315/1138/1/012042

Sun, H., Yang, L., Gu, Y., Pan, J., Wan, F., and Song, C. (2023). Bridging locomotion and manipulation using reconfigurable robotic limbs via reinforcement learning. *Biomimetics* 8, 364. doi:10.3390/biomimetics8040364

Tokekar, P., Vander Hook, J., Mulla, D., and Isler, V. (2016). Sensor planning for a symbiotic uav and ugv system for precision agriculture. *IEEE Trans. robotics* 32, 1498–1511. doi:10.1109/tro.2016.2603528

Torabi, F., Warnell, G., and Stone, P. (2018). Generative adversarial imitation from observation. *arXiv preprint arXiv:1807.06158*

Torabi, F., Warnell, G., and Stone, P. (2019). Recent advances in imitation learning from observation. 6325, 6331. doi:10.24963/ijcai.2019/882*arXiv preprint arXiv:1905.13566*

Tsai, C.-Y., Chou, Y.-S., Wong, C.-C., Lai, Y.-C., and Huang, C.-C. (2019). Visually guided picking control of an omnidirectional mobile manipulator based on end-to-end multi-task imitation learning. *IEEE Access* 8, 1882–1891. doi:10.1109/access.2019.2962335

Tsai, C.-Y., Huang, C.-C., and Chou, Y.-S. (2018). "Data-driven visual picking control of a 6-dof manipulator using end-to-end imitation learning," in 2018 international automatic control conference (CACS) (IEEE), 1–6.

Urain, J., Funk, N., Peters, J., and Chalvatzaki, G. (2023). "Se (3)-diffusionfields: learning smooth cost functions for joint grasp and motion optimization through diffusion," in 2023 IEEE international conference on robotics and automation (ICRA) (IEEE), 5923–5930.

Wakchaure, M., Patle, B., and Mahindrakar, A. (2023). Application of ai techniques and robotics in agriculture: a review. *Artif. Intell. Life Sci.* 3, 100057. doi:10.1016/j.ailsci.2023.100057

[Dataset Wang, P. (2019). On defining artificial intelligence. J. Artif. Gen. Intell. 10 (2), 1–37. doi:10.2478/jagi-2019-0002

Wang, R., Ciliberto, C., Amadori, P., and Demiris, Y. (2020). Support-weighted adversarial imitation learning. *arXiv preprint arXiv:2002.08803*

Wolfert, S., Ge, L., Verdouw, C., and Bogaardt, M.-J. (2017). Big data in smart farming-a review. Agric. Syst. 153, 69-80. doi:10.1016/j.agsy.2017.01.023

Wong, J., Tung, A., Kurenkov, A., Mandlekar, A., Fei-Fei, L., Savarese, S., et al. (2022). "Error-aware imitation learning from teleoperation data for mobile manipulation," in *Conference on robot learning* (PMLR), 1367–1378.

Wu, Y.-H., Charoenphakdee, N., Bao, H., Tangkaratt, V., and Sugiyama, M. (2019). "Imitation learning from imperfect demonstration," in *International conference on machine learning* (PMLR), 6818–6827.

Wu, Z., Chen, Y., Zhao, B., Kang, X., and Ding, Y. (2021). Review of weed detection methods based on computer vision. *Sensors* 21, 3647. doi:10.3390/s21113647

Yamin, N., and Bhat, G. (2022). Near-optimal energy management for energy harvesting iot devices using imitation learning. *IEEE Trans. Computer-Aided Des. Integr. Circuits Syst.* 41, 4551–4562. doi:10.1109/tcad.2022.3198909

Yang, C., Liang, P., Ajoudani, A., Li, Z., and Bicchi, A. (2016). "Development of a robotic teaching interface for human to human skill transfer," in 2016 IEEE/RSJ international Conference on intelligent Robots and systems (IROS) (IEEE), 710–716.

Yang, K., Liu, X., Liu, C., and Wang, Z. (2023). Motion-control strategy for a heavyduty transport hexapod robot on rugged agricultural terrains. *Agriculture* 13, 2131. doi:10.3390/agriculture13112131

Yang, M., Levine, S., and Nachum, O. (2021). Trail: near-optimal imitation learning with suboptimal data. *arXiv preprint arXiv:2110.14770*

Yépez-Ponce, D. F., Salcedo, J. V., Rosero-Montalvo, P. D., and Sanchis, J. (2023). Mobile robotics in smart farming: current trends and applications. *Front. Artif. Intell.* 6, 1213330. doi:10.3389/frai.2023.1213330

Yu, X., Lyu, Y., and Tsang, I. (2020). "Intrinsic reward driven imitation learning via generative model," in *International conference on machine learning* (PMLR, July 18-24, 2021), 10925–10935.

Zare, M., Kebria, P. M., Khosravi, A., and Nahavandi, S. (2024). A survey of imitation learning: algorithms, recent developments, and challenges. *IEEE Trans. Cybern.*, 1–14. doi:10.1109/tcyb.2024.3395626

Zeng, A., Florence, P., Tompson, J., Welker, S., Chien, J., Attarian, M., et al. (2021). "Transporter networks: rearranging the visual world for robotic manipulation," in *Conference on robot learning (PMLR)*, 726–747.

Zhang, A., McAllister, R., Calandra, R., Gal, Y., and Levine, S. (2020). Learning invariant representations for reinforcement learning without reconstruction. *arXiv Prepr. arXiv:2006.* doi:10.48550/arXiv.2006.10742

Zhang, C., Cai, P., Fu, Y., Yuan, H., and Lu, Z. (2023a). Creative agents: empowering agents with imagination for creative tasks. *arXiv Prepr. arXiv:2312.02519*. doi:10.48550/arXiv.2312.02519

Zhang, J., and Cho, K. (2017). Query-efficient imitation learning for end-to-end simulated driving. Proc. AAAI Conf. Artif. Intell. 31. doi:10.1609/aaai.v31i1.10857

Zhang, W., Xu, H., Niu, H., Cheng, P., Li, M., Zhang, H., et al. (2023b). "Discriminator-guided model-based offline imitation learning," in *Conference on robot learning* (PMLR), 1266–1276.

Zhao, J., Yang, Y., Zheng, H., and Dong, Y. (2020). Global agricultural robotics research and development: trend forecasts. J. Phys. Conf. Ser. 1693, 012227. doi:10.1088/1742-6596/1693/1/012227

Zhou, H., Wang, X., Au, W., Kang, H., and Chen, C. (2022a). Intelligent robots for fruit harvesting: recent developments and future challenges. *Precis. Agric.* 23, 1856–1907. doi:10.1007/s11119-022-09913-3

Zhou, K., Liu, Z., Qiao, Y., Xiang, T., and Loy, C. C. (2022b). Domain generalization: a survey. *IEEE Trans. Pattern Analysis Mach. Intell.* 45, 4396–4415. doi:10.1109/tpami.2022.3195549